UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(D) OF THE SECURITIES EXCHANGE ACT OF 1934

Date of Report (Date of earliest event reported): February 11, 2021

Red Cat Holdings, Inc.

(Exact name of registrant as specified in its charter)

<u>Nevada</u> (State or other jurisdiction of incorporation) 000-31587 (Commission

File Number)

<u>86-0490034</u> (I.R.S. Employer

(I.R.S. Employer Identification No.)

370 Harbour Drive Palmas del Mar Humacao, PR 00791

(Address of principal executive offices) (zip code)

(833) 373-3228

(Registrant's telephone number, including area code)

(Former Name or Former Address, if Changed Since Last Report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (*see* General Instruction A.2. below):

□ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)

□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)

□ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))

□ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
None		

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company \Box

L

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Item 1.01. Entry into a Material Definitive Agreement.

On February 11, 2021, Red Cat Holdings, Inc., a Nevada corporation (the "<u>Company</u>") and Red Cat Skypersonic, Inc. Inc., a Nevada corporation ("<u>Acquisition</u>") entered into Share Purchase and Liquidity Event Agreements (the "<u>Agreements</u>") with Giuseppe Santangelo the founder and majority shareholder of Skypersonic, Inc., a Michigan corporation ("<u>Skypersonic</u>") and the holders of common stock and SAFE agreements representing 97.46% of Skypersonic (the "<u>Sellers</u>"), pursuant to which, subject to the satisfaction of certain closing conditions, Acquisition will acquire all of the issued and outstanding share capital of Skypersonic in consideration for an aggregate of Three Million Dollars (\$3,000,000) in shares (the "<u>Share Consideration</u>") of Company common stock, par value \$0.001 per share, (the "<u>Common Stock</u>") at an agreed upon value based upon the VWAP of the Company at closing of the transaction. Fifty (50%) percent of the Share Consideration (the "<u>Escrow Shares</u>") is required to be deposited in an escrow account pursuant to the Agreement for a period of twelve (12) months as security for indemnification on a dollar of dollar basis. The Company agreed that in the event that within 12 months following closing of the Acquisition, the Company issues Common Stock for a price per share less than \$2.50 per share in a Qualified Offering (defined as a public offering or equity or convertible securities in which the Company raises a minimum of \$2 million), the Company shall issue Sellers additional shares of Common Stock equal to the difference between the number of shares issued and the quotient of the Purchase Price divided by the Qualified Offering

Price. Santangelo's and certain principal shareholders have agreed to indemnification obligations, on a pro-rata basis, subject to certain limitations, which shall survive for a period of eighteen (18) months following closing, and includes a basket amount of Twenty-Five Thousand Dollars (\$25,000) before any claim can be asserted and a cap equal to the value of the Escrow Shares or the Share Consideration.

For a period of three (3) years following Closing, Santangelo shall not engage in a business competing with or providing products, services or solutions to the drone industry, first person view ("<u>FPV</u>") business, navigation and software solutions that provide analytics, storage or services for or in conjunction with the drone industry.

On February 15, 2021 the Company's Fat Shark Holdings, Ltd. subsidiary agreed to provide a short-term bridge loan to Skypersonic in the amount of \$75,000 under a Senior Secured Promissory Note due May 14, 2021 (the "Bridge Loan"). Advances under the Bridge Loan accrue interest at a rate of six (6%) percent per annum. The Bridge Loan is secured by shares of Skypersonic common stock pledged as collateral.

The closing of the Skypersonic Transaction is subject to customary closing conditions and is expected to close on or before May 14, 2021.

The Company does not deem the Skypersonic financial condition and results of operations material to the overall financial condition and results of operations of the Company on a consolidated basis and accordingly will not be required to report, in compliance with Regulation S-X (17 CFR 210.8-05), or file audited and pro forma financial statements within 71 calendar days of the date of this Current Report on Form 8-K.

The foregoing descriptions of the terms of the Share Purchase Agreement. Liquidity Event Agreement and Bridge Loan are qualified in their entirety by reference to the full text of the agreements filed as Exhibit 10.1, Exhibit 10.2 and Exhibit 10.3 to this Current Report on Form 8-K.

Following closing of the Transaction, we will continue to be a "smaller reporting company," as defined in Item 10(f)(1) of Regulation S-K, as promulgated by the SEC.

Item 7.01 Regulation FD Disclosure.

On February 16, 2021 the Company issued a press release titled "*Red Cat to Acquire Skypersonic*". A copy of the press release is included herewith as Exhibit 99.1.

On February 16, 2021 the Company posted on its website a presentation titled "*RED CAT HOLDINGS – Investing in the future of drones*". A copy of the presentation is included herewith as Exhibit 99.2.

Item 9.01. Financial Statements and Exhibits.

In accordance with General Instruction B.2 of Form 8-K, the information in this Current Report on Form 8-K under Item 7.01, including Exhibit 99.1 and 99.2 hereto, shall not be deemed to be "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended, or otherwise subject to the liabilities of that section, and shall not be deemed to be incorporated by reference into any of the Company's filings under the Securities Act of 1933, as amended, or the Securities Exchange Act of 1934, as amended, whether made before or after the date hereof and regardless of any general incorporation language in such filings, except to the extent expressly set forth by specific reference in such a filing.

(d) Exhibits

Exhibit Number.	Description
Exhibit 10.1	Share Purchase Agreement dated February 11, 2021.
Exhibit 10.2	Form of Liquidity Event Agreement.
Exhibit 10.3	Senior Secured Promissory Note dated February 15, 2021.
Exhibit 99.1	Press Release dated February 16, 2021.
Exhibit 99.2	Investor Presentation dated February 16, 2021.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Date: February 17, 2021

RED CAT HOLDINGS, INC.

By: Name: Title: /s/ Jeffrey M. Thompson

Jeffrey M. Thompson President and Chief Executive Officer

SHARE PURCHASE AGREEMENT

This Share Purchase Agreement (this "<u>Agreement</u>") is entered into as of the Effective Date by and among Red Cat Holdings, Inc., a Nevada corporation ("<u>Parent</u>"), Red Cat Skypersonic, Inc., a Nevada corporation ("<u>Buyer</u>"), Skypersonic, Inc., a Michigan corporation ("<u>Company</u>" or "<u>Skypersonic</u>"), Giuseppe Santangelo ("<u>Santangelo</u>"), Anna Piccione ("<u>Piccione</u>"), and Michele De Gregorio ("<u>Gregorio</u>") (Santangelo, Piccione and De Gregorio are individually referred to herein as a "Seller" and collectively as the "<u>Sellers</u>", and Sellers, the Company, Parent and Buyer are sometimes referred to herein individually as a "Party" and collectively as the "<u>Parties</u>").

Preliminary Statements

- A. Skypersonic designs and develops remote piloting software and drone hardware.
- A. Sellers collectively own 99,750 of the 102,350 shares of the issued and outstanding shares of capital stock of Skypersonic. Wayne State University on behalf of its Anderson Engineering Ventures Institute ("<u>WSU</u>") owns the remaining 2,600 shares of the issued and outstanding shares of capital stock of Skypersonic.

Agreement

NOW, THEREFORE, in consideration of the preliminary statements and the respective mutual covenants, representations and warranties contained in this Agreement, the parties agree as set forth below.

ARTICLE 1

DEFINITIONS

In addition to terms defined elsewhere in this Agreement, the following terms when used in this Agreement shall have the meanings indicated below:

"<u>Affiliate</u>" of a specified Person means a Person who (at the time when the determination is to be made) directly or indirectly through one or more intermediaries, controls, or is controlled by, or is under common control with the specified Person. As used in the foregoing sentence, the term "<u>control</u>" (including, with correlative meaning, the terms "<u>controlling</u>," "<u>controlled by</u>" and "<u>under common control with</u>") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities, by contract or otherwise, or such other relationships as, in fact, constitutes actual control.

"<u>Agreed Aggregate Working Capital</u>" means Working Capital of the Company of negative Three Hundred Thousand and 00/100 Dollars (-\$300,000.00) (such that the aggregate of the current liabilities of the Company exceeds the assets of the Company (including without limitation, cash, cash equivalents, accounts receivable, inventory and prepaid expense) by Three Hundred Thousand and 00/100 Dollars (\$300,000.00)), determined in accordance with GAAP.

"Agreed Parent Share Price" is defined in Section 2.2(b).

"Agreement" is defined in the Preamble and includes all exhibits and schedules referred to herein.

"<u>Applicable Laws</u>" means, in relation to any Person or Persons, applicable Securities Laws and all other statutes, regulations, statutory rules, orders, by-laws, codes, ordinances, decrees, the terms and conditions of any grant of approval, permission, authority or license, or any judgment, order, decision, ruling, award, policy or guideline, of any Governmental Entity that are applicable to such person or persons or its or their business, undertaking, property or securities and emanate from a Governmental Entity, having jurisdiction over such person or persons or its or their business, undertaking, property or securities;

"<u>Approved Exchange</u>" means, with respect to the Stock Consideration, any major securities exchange, the NASDAQ or any other exchange or quotation system maintained by OTC Markets or if not available, any quotation system or service providing regularly published securities prices designated by Parent.

"Basket Amount" is defined in Section 7.5(b).

"Parent" is defined in the Preamble.

"<u>Claim</u>" is defined in Section 7.4(d)(i).

"<u>Closing</u>" is defined in Section 2.1(b).

"Closing Date" means March 1, 2021 or such other closing date as may be mutually agreed to among the parties in writing.

"Closing Shares" is defined in Section 2.2(b)(ii).

"Closing Trial Balance" is defined in Section 2.3(a).

"Code" means the Internal Revenue Code of 1986, as amended.

"<u>Company</u>" is defined in the Preamble.

"Contracts" is defined in Section 5.19(a).

"Corporate Minute Books" is defined in Section 5.6.

"Costs" is defined in Section 10.7.

"Disclosure Schedules" is defined in the Article 5.

"<u>Disposition</u>" is defined in Section 6.4(b).

"<u>Effective Date</u>" means the date on which this Agreement is fully executed by all Parties, as reflected by the date specified on the signature page hereof beneath their respective signatures.

"<u>Environmental Laws</u>" means any applicable statute, law, ordinance, regulation, rule, code or order and any enforceable judicial or administrative interpretation thereof, including any judicial or administrative order, consent decree or judgment, relating to pollution or protection of the environment or natural resources, including, without limitation, those relating to the use, handling, transportation, treatment, storage, disposal, release or discharge of Hazardous Materials.

"Escrow" means all the Escrow Shares deposited with the Escrow Agent under the Escrow Agreement.

"Escrow Agent" means Equity Stock Transfer, LLC, the Parent's SEC registered transfer agent, identified as the escrow agent specified in the Escrow Agreement.

"<u>Escrow Agreement</u>" means the escrow agreement to be entered into as of the Closing Date among the Sellers, the SAFE Holders, Parent, Buyer, and the escrow agent, in substantially the form attached as <u>Exhibit A</u>.

"Escrow Shares" is defined in Section 2.2(b)(ii).

"Estimated Working Capital" is defined in Section 2.3(a).

"Estimated Working Capital Deficiency Amount" is defined in Section 2.3(a).

"Estimated Working Capital Excess Amount" is defined in Section 2.3(a).

"Estimated Working Capital Statement" is defined in Section 2.3(a).

"Financial Statements" is defined in Section 5.9.

"GAAP" means U.S. generally accepted accounting principles, consistently applied.

"<u>Hazardous Materials</u>" means (a) any petroleum, petroleum products, by-products or breakdown products, radioactive materials, asbestos-containing materials or polychlorinated byphenyls or (b) any chemical, material or substance defined or regulated as toxic or hazardous or as a pollutant or contaminant or waste under any applicable Environmental Laws.

"<u>Indemnified Losses</u>" is defined in Section 7.4(a)(i)(A).

"Indemnified Party" is defined in Section 7.4(c).

"Indemnifying Party" is defined in Section 7.4(c).

"<u>Independent Accounting Firm</u>" is defined in Section 2.3(c).

"Intellectual Property" means any or all of the following owned, licensed, used, controlled by or residing in Company prior to the Closing Date: (a) all inventions (whether patentable or unpatentable and whether or not reduced to practice), all improvements thereto, and all patents, patent applications, and patent disclosures, together with all reissuances, continuations, continuations-in-part, revisions, extensions, and reexaminations thereof; (b) all trademarks, service marks, trade dress, logos, trade names, and corporate names, together with all translations, adaptations, derivations, and combinations thereof and including all goodwill associated therewith, and all applications, registrations, and renewals in connection therewith; (c) all copyrightable works, all copyrights in both published and unpublished works, and all applications, registrations, and related documentation); (f) all other property rights and all licenses and sublicenses granted by or to either Company that relate to any of the foregoing; and (g) all copies and tangible embodiments thereof (in whatever form or medium).

"IRS" means the United States Internal Revenue Service.

"Know-How" means all trade secrets and confidential business information (including, without limitation, databases, ideas, research and development, know-how, formulas, compositions, manufacturing and production processes and techniques, technical data, designs, plans, drawings, specifications, blueprints, customer and supplier lists, pricing and cost information, and business and marketing plans and proposals). "Knowledge" means, with respect to any representation or warranty or other statement in this Agreement qualified by knowledge, the actual knowledge, information or belief of any party as to the matters that are the subject of such representation, warranty or other statement, or any knowledge, information, or belief that such party should have after a due and diligent investigation. Where reference is made to the knowledge of the a Party, such reference shall be deemed to include the directors, officers and managerial employees of such Party.

"Law" means any applicable law, statute, ordinance, rule, regulation, order, writ, judgment or decree.

"Liabilities" means any liability (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated and whether due or to become due).

"Liens" means any liens, claims, charges, rights, pledges, security interests, mortgages, options, title defects or other encumbrances, restrictions or limitations of any nature whatsoever.

"Liquidity Event Agreement" is defined in Section 2.1(c),

"<u>Material Adverse Effect</u>" means any change in or effect on the business of either the Company or Parent (as the case may be) that is, or could reasonably be expected to be, materially adverse to the business, assets (including intangible assets), liabilities (contingent or otherwise), condition (financial or otherwise), prospects or results of operations of the Company or Parent.

"<u>Noncompete Period</u>" is defined in Section 6.1(a).

"Parent Common Stock" means the common stock of Parent, par value US\$0.001 per share.

"Organizational Documents" means any and all documents pursuant to which an entity is organized and/or operates under the applicable laws of its jurisdiction.

"Permits" is defined in Section 5.15.

"<u>Person</u>" means any natural person, corporation, company, unincorporated organization, partnership, association, joint stock company, joint venture, trust or government, or any agency or political subdivision of any government, or any other entity.

"<u>Personally Identifiable Information</u>" means any information that alone or in combination with other information held by the Companies, as applicable, can be used to specifically identify a person's name, street address, telephone number, e-mail address, photograph, social insurance number, driver's license number, passport number, credit or debit card number or customer or financial account number or any similar information that is treated as "Personally Identifiable Information" under any Applicable Laws.

"<u>Plans</u>" is defined in Section 5.17(c).

"Principal Sellers" means Santangelo only.

"<u>Product Data</u>" means all manufacturing information and data, and all submissions and correspondence with or to any governmental or regulatory authority regarding any Product, all as any of the above may be in the possession or control of either Company or Sellers.

"<u>Product Inventory</u>" all inventory owned as of the Closing Date by the Company or any Subsidiary (including sample inventory) thereof of finished Product or works in progress or materials used in the manufacture of finished Product, whether held at a location or facility of the Company (or of any other Person on behalf of the Company) or in transit to or from the Company.

"Products" is defined in Section 5.24(a).

"<u>Purchase Price</u>" is defined in Section 2.2(b).

"<u>Purchased Assets</u>" is defined in Section 2.5(a).

"Purchased Shares" is defined in Section 5.7.

"<u>Qualified Offering</u>" means a public offering by Parent of equity or convertible securities in one or more transactions whereby, on a cumulative basis on or prior to the one (1) year anniversary of the Closing Date, a minimum of \$2.0 million in net proceeds is raised by Parent.

"Qualified Offering Price" means the price per share of Common Stock in a Qualified Offering.

"<u>Real Property</u>" is defined in Section 5.14(a).

"<u>Regulatory Approvals</u>" means the applications or approvals of the US Federal Aviation Administration ("<u>FAA</u>") charged with regulating the manufacture, design or operation of unmanned aerial vehicles ("<u>UAV</u>") without a human pilot, which utilize a ground based controller and a system of communications between the two, generally referred to as drones, and all other Permits required by the FAA to conduct the business as it is currently conducted.

"SAFE Agreements" means those SAFE (Simple Agreement for Future Equity) agreements between the Company and each of De

Gregorio, the Anderson Institute of Wayne State University ("<u>WSU</u>"), Luigi Condorelli ("<u>Condorelli</u>"), Cosimo Urbano ("<u>Urbano</u>"), and Anthony Bosco ("<u>Bosco</u>"), pursuant to which each of them is entitled to a share of the proceeds of any sale of the outstanding capital stock of Skypersonic.

"SAFE Holders" means WSU, Condorelli, Urbano, Bosco and De Gregorio as holders of the SAFE Agreements.

"<u>SEC</u>" is defined in Section 3.9.

"SEC Documents" is defined in Section 3.9.

"Securities Act" means the US Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"<u>Sellers</u>" is defined in the Preamble.

"<u>Serious Adverse Event</u>" means, with respect to any Product, any Adverse Experience that results in any of the following outcomes: death, a life-threatening Adverse Experience, hospitalization, a persistent or significant disability or incapacity, or any other effect that may otherwise jeopardize the business as it is currently conducted or may require intervention to prevent one of the aforementioned outcomes.

"<u>Skypersonic</u>" is defined in the Preamble.

"Stock Consideration" is defined in Section 2.2(b).

"<u>Subsidiary</u>" means any entity (whether or not capitalized) in which a Person (other than an individual) owns, directly or indirectly, an equity interest of 50% or more, or any entity which may be controlled, directly or indirectly, by such Person, whether through the ownership of voting securities, by contract, or otherwise.

"Tax(es)" means any federal, state, local or foreign income, gross receipts, franchise, estimated, alternative minimum, add-on minimum, sales, use, transfer, registration, all gross receipts, *ad valorem*, value added, excise, natural resources, severance, stamp, occupation, premium, windfall profit, assets, minimum income, environmental, customs duties, fees, real property, personal property, capital stock, social security obligations or contributions, unemployment, disability, payroll, license, employee or other withholding, or other tax or governmental charge, of any kind whatsoever, including any interest, penalties or additions to tax or additional amounts in respect of the foregoing; the foregoing shall include any transferee or secondary liability for a Tax and any liability assumed by agreement or arising as a result of being (or ceasing to be) a member of any affiliated group (or being included, or required to be included, in any Tax Return relating thereto).

"Tax Returns" means all material Federal, state, foreign and local tax reports, returns, information returns, declarations, statements, and other documents relating to Taxes.

"Transaction Documents" means this Agreement, the Escrow Agreement and all other documents to be executed and delivered by either party pursuant to or in connection with this Agreement and consummation of the transactions contemplated hereby.

"Transfer Taxes" is defined in Section 6.6(i).

"<u>Trading</u>" is defined in the Preamble.

"<u>VWAP</u>" means the average of the Daily VWAP for the twenty (20) trading days ending on and including the Closing Date (such period, the "<u>20 Day VWAP Period</u>"). "<u>Daily VWAP</u>" means, for any trading day, the per share volume-weighted average price of Parent's common stock as displayed on Bloomberg, L.P. (or its equivalent successor if such service is not available) in respect of the period from the scheduled open of trading until the scheduled close of trading of the primary trading session on such trading day on the Nasdaq Capital Market or, if unavailable, on the OTC Markets. The Daily VWAP will be determined without regard to after-hours trading or any other trading outside of the regular trading session trading hours.

"<u>Working Capital</u>" means current assets of the Company (including without limitation, cash, cash equivalents, accounts receivable, inventory and prepaid expense) less current liabilities of the Company, determined in accordance with GAAP.

ARTICLE 2

PURCHASE OF SECURITIES; CONSIDERATION

2.1 Securities to be Purchased; Closing.

(a) Subject to the terms and conditions set forth herein, on the Closing Date, Sellers shall sell and transfer to Parent, and Parent shall purchase from Sellers, all of their respective right, title and interest in and to the Purchased Shares, free and clear of all Liens, which Purchased Shares shall in the aggregate represent ninety-seven and 46/100 percent (97.46%) (rounded) of the issued and outstanding shares of Skypersonic as of the Closing Date, exclusive of the shares held by WSU and its SAFE Agreement addressed in Section 2.2(a) hereof.

(b) Subject to the terms and conditions of this Agreement, the sale and purchase of the Purchased Shares contemplated hereby and the transfers and deliveries to be made pursuant to this Agreement shall take place at a closing on the Closing Date at the offices of Parent at 370 Harbour Drive, Palmas del Mar, Humacao, Puerto Rico, 00791 (the "<u>Closing</u>") at 12:00 p.m. local time, at such other place as may be agreed to

by the parties in writing, or through an escrow with a mutually acceptable closing escrow agent where the Parties and their respective attorneys need not be physically present and may deliver documents by overnight courier or other means. All proceedings to be taken and all documents to be executed at the Closing shall be deemed to have been taken, delivered and executed simultaneously, and no proceeding shall be deemed taken nor documents deemed executed or delivered until all have been taken, delivered and executed.

(c) At the Closing, the Sellers shall deliver or cause to be delivered to Parent (i) executed share transfer instruments relating to One Hundred Percent (100%) of the Purchased Shares of Skypersonic, sufficient to convey to Parent good title to the Purchased Shares of Skypersonic, free and clear of any and all claims or Liens of any nature whatsoever and together with all accrued benefits and rights attaching thereto, (ii) a written agreement by each SAFE Holder (other than WSU) by which such SAFE Holder agrees to accept a portion of the Share Consideration in full satisfaction and discharge of such SAFE Holder's SAFE Agreement (each, a "Liquidity Event Agreement"); (iii) such other documents as may be specified, or required to satisfy the conditions within Sellers' control set forth, in Sections 9.1 and 9.2, and (iv) the Sellers shall cause a board meeting of Skypersonic approving the transfer of the Purchased Shares, the updating of the share register of Skypersonic, appointment of representatives of Parent to such offices and as directors of Skypersonic as requested by Parent, with such authorizations (banking and otherwise) as required by Parent, and any other ancillary matters.

2.2 Closing Consideration.

(a) Subject to any applicable adjustments set forth in Section 2.3 or Section 2.5 hereof (each, a "Price Adjustment"), the total consideration to be paid for one hundred percent (100%) of the issued and outstanding capital stock of the Company (including the 2,600 shares held by WSU) and for the full satisfaction of all SAFE Agreements (including the SAFE Agreement held by WSU) shall be Three Million Dollars (US\$3,000,000), payable in shares of Parent Common Stock at an agreed upon value based upon the VWAP (the "Aggregate Purchase Price"). The Parties acknowledge that WSU is not a party to this Agreement concerning its shares of the common stock of Skypersonic and has not yet executed and delivered a Liquidity Event Agreement concerning its SAFE Agreement. The portion of the Aggregate Purchase Price attributable to the 2,600 shares of Skypersonic stock held by WSU is \$52,819.25 and the portion of the Aggregate Purchase Price attributable to the satisfaction and discharge of WSU's SAFE Agreement is \$375,000 (in each case subject to pro rata adjustment on account of such a separate purchase agreement for the sale of WSU's shares of Skypersonic common stock for \$52,819.25 of the Aggregate Purchase Price and WSU's delivery of a Liquidity Event Agreement for the full satisfaction and discharge of its SAFE Agreement for the full satisfaction and discharge of its SAFE Agreement for the full satisfaction and discharge of its SAFE Agreement for the full satisfaction and discharge of the common stock of Skypersonic common stock for \$52,819.25 of the Aggregate Purchase Price and WSU's delivery of a Liquidity Event Agreement for the full satisfaction and discharge of its SAFE Agreement for \$375,000 of the Aggregate Purchase Price (in each case subject to pro rata adjustment on account of any Price Adjustment by WSU only upon a separate closing on the purchase of WSU's shares of Skypersonic common stock and upon delivery of a Liquidity Event Agreement for the full satisfaction and discharge of its SAFE Agreement for the sa

(b) Subject to any applicable Price Adjustment, in consideration of the sale, assignment, transfer and delivery of the Purchased Shares by the Sellers to Buyer and the Liquidity Event Agreements by the SAFE Holders (other than WSU), at the Closing, Parent shall deliver Two Million Five Hundred Seventy-Two Thousand One Hundred Eighty and 75/100 Dollars (\$2,572,180.75), payable in shares of Parent Common Stock (the "Stock Consideration") at an agreed upon value (the "Agreed Parent Share Price") based upon the VWAP (the "Purchase Price"), to be issued to the Sellers and SAFE Holders at Closing and delivered as follows:

(i) delivery to the Sellers and SAFE Holders of an aggregate of fifty (50%) percent of the Stock Consideration (the "Closing Shares"); and

(ii) delivery to the Escrow Agent of the remaining fifty (50%) percent of the Stock Consideration (the "Escrow Shares").

(c) One Hundred (100%) Percent of the total Stock Consideration shall be issued at Closing and Sellers and SAFE Holders shall have all rights of a shareholder with respect to all of the Closing Shares and the Escrow Shares, including rights to vote and receive dividends or other distributions on account of such shares. The Closing Shares and the Escrow Shares (upon release under the terms of the Escrow Agreement) shall be distributed to Sellers and the SAFE Holders in such proportions as Skypersonic and Sellers shall jointly specify at Closing.

2.3 Estimated Working Capital.

(a) No more than ten and no later than three days prior to the Closing Date, the Sellers shall cause the Company to prepare and deliver to Parent (i) an unaudited preliminary balance sheet of the Company as of the Closing Date and a preliminary unaudited profit and loss statement for the period ending on the Closing Date, prepared in accordance with the same accounting method and practices utilized in the preparation of the Financial Statements (collectively, the "<u>Closing Trial Balance</u>"), and (ii) a statement (the "<u>Estimated Working Capital</u> <u>Statement</u>") setting forth the Company's reasonable and good faith estimate of the Working Capital of the Company as of the Closing Date ("<u>Closing Date Working Capital</u>") and the components and calculations thereof in reasonable detail, by reference to the foregoing Closing Trial Balance.

(b) If the Closing Date Working Capital reflected on the Estimated Working Capital Statement is less than the Agreed Aggregate Working Capital (such that the aggregate of the current liabilities of the Company exceeds the assets of the Company (including without limitation, cash, cash equivalents, accounts receivable, inventory and prepaid expense) by more than Three Hundred Thousand and 00/100 Dollars (\$300,000.00)), determined in accordance with GAAP), the Purchase Price shall be reduced on a dollar-for-dollar basis by the amount by which the Agreed Aggregate Working Capital exceeds the Closing Date Working Capital.

(c) If the Closing Date Working Capital reflected on the Estimated Working Capital Statement is greater than the Agreed Aggregate Working Capital (such that the aggregate of the current liabilities of the Company exceeds the assets of the Company (including without

limitation, cash, cash equivalents, accounts receivable, inventory and prepaid expense) by less than Three Hundred Thousand and 00/100 Dollars (\$300,000.00)), determined in accordance with GAAP), the Purchase Price shall be increased on a dollar-for-dollar basis by the amount by which the Closing Date Working Capital exceeds the Agreed Aggregate Working Capital.

2.4 Escrow Shares.

(a) The Escrow shall provide security for the Sellers' payments to Buyer of all amounts due as a result of the indemnification obligations in Section 7.4(a)(i)(A) and Section 7.4(b)(i).

(b) Any Claim against the Escrow must be made within twelve (12) months following Closing.

(c) Amounts to be satisfied from the Escrow pursuant to Section 2.4(a) will be satisfied by the cancellation of Escrow Shares having a value equal to the amount to be satisfied pursuant to Section 2.4(a), with such Escrow Shares valued at the Agreed Parent Share Price. All Escrow Shares remaining after any such cancellation of shall, upon release from Escrow pursuant to the terms of the Escrow Agreement, be release to the Sellers and SAFE Holders in such proportions as Sellers and SAFE Holders may specify by written notice to Parent and Escrow Agrent.

(d) Except to the extent of the limitations provide in Section 7.5, the Sellers' liability to Buyer for amounts due as a result of the indemnification obligations in Article 7 shall be limited to amounts or assets in the Escrow.

2.5 <u>Top-Up/True-Up Issuance of Parent Stock</u>. It is the Parties' intention that the Purchase Price will be payable by issuance to Sellers of a number of newly issued shares of Parent Common Stock equal to the quotient of (a) Three Million and 00/100 Dollars (\$3,000,000.00) divided by (b) the lower of (i) the VWAP and (ii) the Qualified Offering Price. In the event it is determined within twelve (12) months after the Closing that the Qualified Offering Price of Parent Common Stock issued in a Qualified Offering is less than the VWAP, Parent shall issue to Sellers additional shares of Parent Common Stock equal to the difference between the number of shares of Parent Common Stock issued to Sellers at Closing and the number of Shares of Common Stock equal to the quotient of the Purchase Price divided by the Qualified Offering Price.

ARTICLE 3

REPRESENTATIONS AND WARRANTIES REGARDING PARENT

In order to induce the Sellers to enter into this Agreement and to consummate the transactions contemplated hereby, Parent and Buyer jointly and severally make the representations and warranties set forth below to Sellers on the date hereof, except as set forth in the SEC Documents if reasonably apparent that any event, item or occurrence disclosed in such SEC Documents is an event, item or occurrence that relates to a matter covered by any representation or warranty set forth in this Article 3) or in a Disclosure Schedule delivered by Parent to the Sellers (the "Parent Disclosure Schedule").

3.1 <u>Organization</u>. Parent is a corporation duly organized and validly existing under the laws of the State of Nevada. Buyer is a corporation duly organized and validly existing under the laws of the State of Nevada.

3.2 <u>Authorization; Enforceability</u>. Parent and Buyer each has all requisite right, power and authority to execute and deliver the Transaction Documents and to consummate the transactions contemplated thereby. The execution and delivery of the Transaction Documents by Parent and Buyer and the consummation by Parent and Buyer of the transactions contemplated thereby have been duly authorized by all requisite corporate action. This Agreement has been duly executed and delivered by Parent and Buyer and constitutes the legal, valid and binding obligation of Parent and Buyer, enforceable against each in accordance with its terms, except as such enforceability may be limited by applicable bankruptcy, insolvency, reorganization, moratorium and other similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity, whether considered in a proceeding in equity or at law.

3.3 <u>No Consent, Violation or Conflict</u>. The execution and delivery of the Transaction Documents by Parent and Buyer, the consummation by Parent and Buyer of the transactions contemplated thereby, and compliance by each with the provisions hereof: (a) do not and will not violate or, if applicable, conflict with any provision of Law, or any provision of the Organizational Documents of Parent or Buyer; and (b) do not and will not, with or without the passage of time or the giving of notice, result in the breach of, cause the acceleration of performance or constitute a default or require any consent under, any instrument or agreement to which Parent or Buyer is a party or by which Parent or Buyer or any of their respective properties may be bound or affected, other than instruments or agreements as to which consent shall have been obtained at or prior to the Closing Date or any breaches or defaults which would not affect Parent's ability to consummate the transactions contemplated thereby.

3.4 <u>Brokers</u>. Neither Parent nor Buyer has employed any financial advisor, broker or finder, and neither Parent nor Buyer has incurred and will not incur any broker's, finder's, investment banking or similar fees, commissions or expenses, in connection with the transactions contemplated by this Agreement.

3.5 <u>Consent of Governmental Authorities</u>. Parent and Buyer have each obtained, or will have obtained prior to the Closing, all necessary authorizations to enter into this Agreement and any other Transaction Document to which each is a party, to perform their respective obligations hereunder and thereunder, and to consummate the transaction contemplated hereby and thereby. No further consent, approval or authorization of, or registration, qualification or filing with any governmental or regulatory authority, or any other Person, is required to be made or obtained by Parent or Buyer in connection with the execution and delivery of the Transaction Documents by Parent or Buyer, or the consummation by Parent or Buyer of the transactions contemplated thereby.

3.6 <u>Legal Proceedings</u>. There is no action, claim, dispute, suit, investigation or proceeding pending or, to Parent's and Buyer's Knowledge, threatened against Parent, Buyer or any of their respective properties or rights, nor any judgment, order, injunction or decree before any court, arbitrator or administrative or governmental body which might adversely affect or restrict the ability of Parent or Buyer to consummate the transactions contemplated by the Transaction Documents, or to perform their respective obligations thereunder.

3.7<u>Knowledge</u>. Parent and Buyer each has sufficient knowledge, experience and sophistication in financial and business matters to be capable of evaluating the merits and risks of the purchase of and investment in the Purchased Shares and of making an informed investment decision with respect thereto. Parent and Buyer further confirm that the Closing shall indicate that each of Parent, Buyer and their respective attorneys, accountants and other advisors: (a) were given the opportunity to examine the financial and business situation of Skypersonic, and to ask questions and receive answers from the management of Skypersonic; (b) obtained from the Company and Sellers all information requested; and (c) obtained sufficient information to evaluate the merits and risks of an investment in the Purchased Shares.

3.8 <u>Validity of Stock Consideration</u>. When issued and delivered to Sellers and Escrow Agent in accordance with this Agreement, the Stock Consideration shall be (a) duly and validly authorized, issued and outstanding, fully paid and non-assessable and (b) free and clear of any Liens, except as provided in the escrow provisions specified in this Agreement.

3.9 <u>SEC Documents and Compliance</u>. Since September 1, 2019, Parent has filed all reports, schedules, forms, statements and other documents required to be filed by it with the U.S. Securities and Exchange Commission ("SEC"), pursuant to the reporting requirements of all applicable federal and securities laws (all of the foregoing that were scheduled to be filed prior to the Closing Date, and all exhibits included therein and financial statements and schedules thereto and documents incorporated by reference therein, are hereinafter referred to as the "SEC Documents"). The SEC Documents comply in all material respects with the requirements of all applicable federal securities laws, rules and regulations. The SEC Documents do not contain any untrue statement of a material fact of omit to state a material fact required to be stated therein or necessary in order to make the statements therein, in the light of the circumstances under which they were made, not misleading.

ARTICLE 4

REPRESENTATIONS AND WARRANTIES REGARDING THE SELLERS

In order to induce Parent any Buyer to enter into this Agreement and to consummate the transactions contemplated hereby, each of the Sellers (individually as to themselves and their respective shares of Purchased Stock only and not jointly and severally) make the representations and warranties set forth below to Parent as of the date hereof. For avoidance of doubt, so Seller makes any representation or warranty as to any other Seller or the shares of Purchased Stock held by any other Seller.

4.1 <u>Authorization; Enforceability</u>. The Seller has all requisite right, power and capacity to execute and deliver the Transaction Documents to which it is a party and consummate the transactions contemplated thereby. The execution and delivery of the Transaction Documents by a Seller to which it is a party and the consummation by such Seller of the transactions contemplated thereby have been duly authorized by all requisite action. This Agreement has been duly executed and delivered by Sellers and constitutes the legal, valid and binding obligations of such Party, enforceable against such Seller in accordance with its terms except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity, whether considered in a proceeding in equity or at law.

4.2 <u>No Consent, Violation or Conflict</u>. The execution and delivery of the Transaction Documents by a Seller to which it is a party, the consummation of the transactions contemplated thereby, and compliance by the Seller with the provisions hereof, (a) do not and will not violate or, if applicable, conflict with any provision of Law and (b) do not and will not, with or without the passage of time or the giving of notice, result in the breach of, cause the acceleration of performance or constitute a default or require any consent under, any instrument or agreement to which such Seller is a party or by which such Sellers' Purchased Shares may be bound or affected.

4.3 <u>Consent of Governmental Authorities</u>. The Seller has obtained all necessary authorizations and no further consent, approval or authorization of, or registration, qualification or filing, with any governmental or regulatory authority, or any other Person, is required to be made or obtained by the Seller in connection with the execution and delivery of the Transaction Documents to which such Seller is a party or the consummation by such Seller of the transactions contemplated thereby.

4.4 <u>Brokers</u>. The Seller has not incurred and will not incur any broker's, finder's, investment banking or similar fees, commissions or expenses in connection with the transactions contemplated by this Agreement, which would be payable by Parent or Skypersonic after the Closing Date.

4.5 <u>Capitalization</u>. The Seller is and will immediately prior to the Closing be the legal, record and beneficial owner of such Seller's shares of the capital stock of Skypersonic included in the Purchased Shares, and such Purchased Shares are and will immediately prior to the Closing be owned free and clear of any Liens whatsoever, including, without limitation, claims or rights under any voting trust agreements, proxies, shareholder agreements or other agreements. At the Closing, the Seller will transfer and convey to Buyer, and Buyer will acquire, good and valid title to such Purchased Shares free and clear of all Liens. No written or oral agreement or understanding with respect to the disposition of Purchased Shares held by the Seller or any rights therein, other than this Agreement, exists.

4.6 <u>**Rights, Warrants, Options.**</u> Other than this Agreement, there are no options, warrants or other rights, arrangements or commitments of any character to which the Seller is a party or by which the Sellers is bound relating to the Purchased Shares held by such Seller or obligating the Seller to sell any shares, or other equity interests in, Skypersonic.

4.7 <u>Legal Proceedings</u>. There are no suits or proceedings pending or threatened before any court or by or before any governmental or regulatory authority, commission, or agency or public regulatory body against the Seller which, if adversely determined, would interfere with

Parent's, Buyer's or such Seller's ability to consummate the transactions contemplated hereby.

4.8 <u>Related Parties</u>. Except as disclosed in <u>Schedule 4.8</u>, neither the Seller, nor any of such Seller's spouse or family has, directly or indirectly, (a) any ownership interest in, or is a director, officer, employee, consultant or agent of, any Person which is a competitor, supplier or customer of the Company; (b) any ownership interest in any property or asset, tangible or intangible, including any Intellectual Property, used in the conduct of the Company's business; (c) any contractual or other arrangement with any competitor, supplier or customer of the Company; or (d) any cause of action or claim whatsoever against, or owes any amount to, the Company.

ARTICLE 5

REPRESENTATIONS AND WARRANTIES OF SELLERS REGARDING SKYPERSONIC

In order to induce Parent and Buyer to enter into this Agreement and to consummate the transactions contemplated hereby, the Principal Sellers make the representations and warranties set forth below to Parent as of the date hereof, except as otherwise noted herein or as set forth in the disclosure schedules to be provided by Principal Sellers pursuant to Section 9.1(m) and Section 9.2(p) hereof (the "Disclosure Schedules"). The obligation of Principal Sellers for the representations and warranties in this Article 5 and the under Section 7.4(a) shall be the in proportion which the number of Purchased Shares held by each Principal Seller bears to the number of Purchased Shares held by all Principal Sellers. All information set forth in the Disclosures Schedules shall clearly identify the specific paragraph to which it applies.

5.1 <u>Organization</u>. Skypersonic is a corporation duly organized and validly existing under the laws of the state of Michigan. Skypersonic is duly qualified or licensed to do business in each jurisdiction where the character of the properties owned or operated by it or the nature of its business makes such qualification or licensing necessary. Skypersonic has all requisite right, power and authority to (a) own and operate its properties, (b) conduct its business as presently conducted in accordance with local business practices and (c) engage in and consummate the transactions contemplated hereby. Skypersonic is not in default under its Organizational Documents.

5.2 <u>Authorization; Enforceability</u>. Skypersonic has all requisite right, power and authority to execute and deliver the Transaction Documents to which it is a party and consummate the transactions contemplated thereby. The execution and delivery by Skypersonic of the Transaction Documents to which it is a party and the consummation by Skypersonic of the transactions contemplated thereby have been duly authorized by all requisite corporate or other action. This Agreement has been duly executed and delivered by Skypersonic and constitutes the legal, valid and binding obligation of Skypersonic, enforceable in accordance with its terms except as such enforceability may be limited by applicable bankruptcy, insolvency, fraudulent transfer, reorganization, moratorium and other similar Laws affecting creditors' rights generally and subject to the effect of general principles of equity, whether considered in a proceeding in equity or at law.

5.3 No Consent, Violation or Conflict. The execution and delivery by Skypersonic of the Transaction Documents to which it is a party and the consummation of the transactions contemplated thereby, and compliance by Skypersonic with the provisions hereof, (a) do not and will not violate or, if applicable, conflict with any provision of Law, or any provision of such Company Organizational Documents, and (b) except as identified on <u>Schedule 5.3</u>, do not and will not, with or without the passage of time or the giving of notice, result in the breach of, cause the acceleration of performance or constitute a default or require any consent under, or result in the creation of any Lien upon any property or assets of such Company pursuant to any instrument or agreement to which such Company is a party or by which such Company's properties may be bound or affected.

5.4<u>Consent of Governmental Authorities</u>. Skypersonic and each Subsidiary of the Company has obtained, or will have obtained prior to Closing, all necessary authorizations to enter into this Agreement and any other Transaction Document to which it is a party, to perform its obligations hereunder and thereunder, and to consummate the transaction contemplated hereby and thereby. Except as identified in <u>Schedule</u> <u>5.3</u>, no further consent, approval or authorization of, or registration, qualification or filing, with any governmental or regulatory authority, or any other Person, is required to be made or obtained by such Company in connection with the execution and delivery of the Transaction Documents by such Company to which it is a party or the consummation by the Companies of the transactions contemplated thereby.

5.5 <u>Brokers</u>. <u>Skypersonic</u> has not incurred or will incur any broker's, finder's, investment banking or similar fees, commissions or expenses in connection with the transactions contemplated by this Agreement, which would be payable by Parent or such Company after the Closing Date.

5.6<u>Organizational Documents and Corporate Records</u>. A true and complete copy of the Organizational Documents of the Company or any Subsidiary, as amended, will be delivered to Parent on the Closing Date. The minute book of each the Company (the "<u>Corporate Minute Books</u>") will also be delivered to Parent on the Closing Date. All matters requiring the authorization or approval of the board of directors and/or the shareholders of the Company or any Subsidiary have been or, on or before Closing will be, duly and validly authorized and approved by them.

5.7 Capitalization.

(a) The authorized share capital of Skypersonic consists of 1,000,000 shares of common stock ("<u>Skypersonic Common Stock</u>"). As of the Effective Date, 102,350 shares of Skypersonic Common Stock are issued and outstanding, of which Sellers are the record holders of 99,750 shares and WSU is the holder of the remaining 2,600 shares. The 99,750 shares of Skypersonic Common Stock held by Sellers collectively comprise the "<u>Purchased Shares</u>".

(b) All of the Purchased Shares, including all Purchased Shares issued prior to the Closing Date, have been and will be duly authorized and are and will be validly issued, fully paid and nonassessable. The Sellers are and immediately prior to the Closing will, directly or indirectly, be the record owners of One Hundred Percent (100%) of the Purchased Shares. The Purchased Shares represent ninety-seven and 46/100 percent (97.46%) (rounded) of the issued and outstanding shares of Skypersonic (exclusive of the right, if any, of WSU under its SAFE

Agreement to receive Purchased Shares). <u>Schedule 5.7(b)</u> lists all Subsidiaries of Skypersonic and each entity in which Skypersonic holds any equity. None of the Purchased Shares was or will be issued in violation of any law, preemptive right, right of first refusal or similar agreement.

5.8 <u>Rights, Warrants, Options</u>. Except for the SAFE Agreements, there are no options, warrants or other rights, arrangements or commitments of any character to which Skypersonic is a party or by which Skypersonic is bound relating to the issued or unissued shares of Skypersonic or obligating such Company to issue or sell any shares, or other equity interests in, such Company. There are no outstanding obligations of Skypersonic to redeem or otherwise acquire any of the Purchased Shares and there are no outstanding contractual obligations of Skypersonic to provide funds to, or make any investment (in the form of a loan, capital contribution or otherwise) in, any other Person.

5.9 Financial Statements. Schedule 5.9 includes a true and complete copy of the unaudited balance sheet of Skypersonic (and Subsidiaries) as of December 31, 2019 and 2020 and the unaudited consolidated profit and loss statement, statement of cash flow and statement of changes in shareholders' equity of Skypersonic (and Subsidiaries) for the period ending on such dates, certified by such Company's chief executive officer or chief executive officer (collectively, the "Financial Statements"). The Financial Statements: (a) have been prepared in accordance with the books of account and records of Skypersonic (and Subsidiaries); (b) fairly present, and are true, correct and complete statements in all material respects of, the consolidated financial condition of Skypersonic (and Subsidiaries) and the results of its operations at the dates and for the periods specified in those statements; and (c) have been prepared in accordance with GAAP consistently applied with prior periods (other than those departures from GAAP noted in Schedule 5.9). The Closing Trial Balance will (a) be prepared in accordance with the books of account and records of Skypersonic (and Subsidiaries) and delivered on an estimated basis and (b) fairly present, the estimated consolidated financial condition of Skypersonic (and Subsidiaries) and the results of statement in accordance with the books of account and records of Skypersonic (and Subsidiaries) and delivered on an estimated basis and (b) fairly present, the

5.10 <u>Absence of Undisclosed Liabilities</u>. To Principal Sellers' Knowledge, as of the Closing Date, Skypersonic has no debts, claims, Liabilities, commitments or obligations of any nature whatsoever, whether accrued, absolute, contingent or otherwise, other than as provided for in this Agreement, the Disclosure Schedules, or disclosed and accrued for or reserved against in the Financial Statements or in the Closing Trial Balance. There is no basis for assertion against either Company of any such debt, claim, Liability, commitment, obligation or loss. For the avoidance of doubt, the Sellers acknowledges that Buyer is not assuming any such debts, claims, Liabilities, commitments, obligations or losses not disclosed and accrued for or reserved against in the Closing Trial Balance, and Buyer will be indemnified for such pursuant to Section 7.4.</u>

5.11 Compliance with Laws.

(a) Skypersonic and each Subsidiary is in compliance with all Laws applicable to it or its properties or has made all necessary filings to be in compliance with all such Laws, including, without limitation, those relating to (i) the development, testing, manufacture, packaging and labeling of products, (ii) employment, occupational safety and employee health, (iii) building, zoning and land use; and (iv) the Foreign Corrupt Practices Act and all applicable anti-bribery, anti-kickback and related laws and regulations. Neither Skypersonic (nor any Subsidiary) has received notification from any governmental or regulatory authority asserting that it is not in compliance with or has violated any of the Laws which such governmental or regulatory authority enforces, or threatening to revoke any authorization, consent, approval, franchise, license, or permit, and Skypersonic (and each Subsidiary) is not subject to any agreement or consent decree with any governmental or regulatory authority arising out of previously asserted violations.

(b) Skypersonic (and each Subsidiary of the Company) has at all times been in compliance with all legal requirements applicable to it, relating to export control, trade embargoes, and anti-boycott prohibitions. No product sold or services provided by Skypersonic (or any Subsidiary of the Company) during the last five (5) years has been, directly or indirectly, sold to or performed on behalf of any country subject to the restricted list of the Office of Foreign Assets Control Regulations (OFAC), or International Traffic In Arms Regulations (ITAE), or similar restrictions of the United States, including Cuba, Iraq, Iran, Libya or North Korea or customers in such countries.

5.12 Legal Proceedings. Except as identified on <u>Schedule 5.12</u>: (a) Neither Skypersonic nor any Subsidiary of the Company is a party to any pending or to Principal Sellers' Knowledge, threatened legal, administrative or other proceeding, arbitration, mediation, out-of-court settlement negotiation or investigation, and (b) to Principal Sellers' Knowledge, no Person who is or was within the last five years a director or officer of Skypersonic or any Subsidiary of the Company is a party to any pending, or threatened, legal, administrative or other proceeding, arbitration, out-of-court settlement negotiation or investigation or investigation or investigation in their capacity as directors or officers of Skypersonic or such Subsidiary, which adversely affects Skypersonic or such Subsidiary. Neither Skypersonic nor any such Subsidiary is subject to any order, writ, injunction, decree or other judgment of any court. Neither Skypersonic or any Subsidiary of the Company is subject to any decree or other similar judgment of any governmental authority. There are no suits or proceedings pending or threatened before any court or by or before any governmental or regulatory authority, commission, or agency or public regulatory body against Skypersonic or any Subsidiary of the Company which, if adversely determined, would interfere with Parent's, Buyer's or Skypersonic's ability to consummate the transactions contemplated hereby.

5.13 Title to and Condition of Personal Property.

(a) Skypersonic and each Subsidiary of the Company has and will have good and marketable title or leasehold interest to each item of equipment and other personal property worth more than \$10,000 included as an asset in the Closing Trial Balance and/or used in connection with the operation of its business.

(b) Each item of equipment, personal property and asset of Skypersonic and any Subsidiary, included as an asset in the Closing Trial Balance and/or used in connection with the operation of its business shall remain with Skypersonic or such Subsidiary. The parties agree that <u>Schedule 5.13(b)</u> sets forth the full and complete list of all assets owned by Skypersonic and each Subsidiary as of the Closing Date. Except as set forth on such <u>Schedule 5.13(b)</u>, none of the items of equipment, personal property and assets included therein was imported under a temporary import or similar regulatory regime that would restrict the transfer, or would cause either Skypersonic or a Subsidiary to owe

additional Taxes as a result of the transfer, of such property.

5.14 Real Property.

(a) <u>Schedule 5.14(a)</u> sets forth the street address of each parcel of real property leased by Skypersonic or any Subsidiary of the Company (the "<u>Real Property</u>"). Parent has been delivered true and complete copies of all of the lease agreements relating to the Real Property. Skypersonic and each Subsidiary enjoys peaceful and undisturbed possession of the Real Property.

(b) Neither Skypersonic nor any Subsidiary of the Company owns any real property.

5.15 <u>Governmental Authorizations</u>. Skypersonic and each Subsidiary of the Company has all authorizations, consents, approvals, franchises, licenses and permits required under applicable Law for the ownership by Skypersonic of its properties and operation of its business as presently operated (the "<u>Permits</u>"). No suspension, nonrenewal or cancellation of any of the Permits is pending or, to any Principal Sellers' Knowledge, threatened, and there is no reasonable basis therefor. Neither Skypersonic nor any Subsidiary is in default or violation of any Permits.

5.16 <u>Compliance with Environmental Laws</u>. Skypersonic and each Subsidiary of the Company has conducted its business in compliance with all applicable Environmental Laws. Except as set forth in <u>Schedule 5.16</u>, there have been no governmental claims, citations, notices of violation, judgments, decrees or orders issued against either Skypersonic or any Subsidiary for impairment or damage, injury or adverse effect to the environment or public health and there have been no private complaints with respect to any such matters. To Principal Sellers' Knowledge, there is no condition relating to any properties of Skypersonic or any Subsidiary that would require any type of remediation, clean-up, response or other action under applicable Environmental Laws and Skypersonic and each Subsidiary has complied with Environmental Laws in the generation, treatment, storage and disposal of toxic and hazardous substances, as defined under any applicable Environmental Laws.

5.17 Employment Matters.

(a) Except as identified in <u>Schedule 5.17(a)(i)</u>, there are no employment, consulting, severance or indemnification arrangements, arrangements which contain change of control provisions, agreements, or understandings between either Skypersonic or any Subsidiary and any officer, director, consultant or employee. <u>Schedule 5.17(a)(ii)</u> contains the names, job descriptions and annual salary rates and other compensation of all officers, directors, employees and consultants of Skypersonic and each Subsidiary (including compensation paid or payable under the Plans (as hereafter defined)), and a list of all employee policies (written or otherwise), employee manuals or other written statements of rules or policies concerning employment, including working conditions, vacation and sick leave, a complete copy of each of which (or a description, if unwritten) has been delivered to Parent.

(b) Skypersonic and each Subsidiary has complied with all applicable employment Laws, including Laws relating to payroll and related obligations, benefits, and social security and, except as identified in <u>Schedule 5.17(b)</u>, Skypersonic does not have any obligation in respect of any amount due to employees of Skypersonic and each Subsidiary or government agencies, other than normal salary, other fringe benefits and contributions accrued but not payable on the date hereof.

(c) <u>Schedule 5.17(c)</u> sets forth a complete list of all pension, retirement, share purchase, share bonus, share ownership, share option, profit sharing, savings, medical, disability, hospitalization, insurance, deferred compensation, bonus, incentive, welfare or any other material employee benefit plan, policy, agreement, commitment, arrangement or practice currently or previously maintained by Skypersonic and each Subsidiary for any of its directors, officers, consultants, employees or former employees (the "<u>Plans</u>").

Without limiting the generality of Section 5.11, each Plan has been administered in accordance with its terms and applicable Law. With respect to the Plans, (i) no event has occurred and there exists no condition, facts or circumstances, which could give rise to any liability of Skypersonic or any Subsidiary under the terms of such Plans or any applicable Law, (ii) each of Skypersonic and each Subsidiary has paid or accrued all amounts required under applicable Law and any Plan to be paid as a contribution to each Plan through the date hereof, (iii) each of Skypersonic and each Subsidiary of the Company has set aside adequate reserves to meet contributions which are not yet due under any Plan, (iv) the fair market value of the assets of each funded Plan, the liability of each insurer for any Plan funded through insurance or the book reserve established for any Plan, together with accrued contributions, is sufficient to procure or provide for the accrued benefit obligations, as of the Closing Date, with respect to all current and former participants in such Plan according to the actuarial assumptions and valuations most recently used to determine employee contributions to such Plan and no transaction contemplated by this Agreement shall cause such assets or insurance obligations to be less than such benefit obligations, and (v) each Plan required to be registered has been registered and has been maintained in good standing with applicable regulatory authorities.

No Plan has been and, to Principal Sellers' Knowledge, no event has occurred prior to the Effective Date which would cause or require a Plan to be, (i) terminated, (ii) amended in any manner which would directly or indirectly increase the benefits accrued, or which may be accrued, by any participant thereunder or (iii) amended in any manner which would materially increase the cost to either Skypersonic or any Subsidiary or Buyer of maintaining such Plan. No Plan provides retiree medical or retiree insurance benefits to any Person. Except as disclosed or noted in Schedule 5.17(b) or in the Financial Statements, there are no amounts due or owing to any employee of Skypersonic or any Subsidiary of the Company for any accrued salary, remuneration, compensation and/or benefit, including, without limitation, amounts due for accrued vacation, sick leave or commissions.

5.18 <u>Labor Relations</u>. There is no strike or dispute pending or threatened involving any employees of either Skypersonic or any Subsidiary. None of the employees of Skypersonic or any Subsidiary is a member of any labor union and neither Skypersonic or any Subsidiary is a party to, otherwise bound by, or threatened with any labor or collective bargaining agreement. None of the employees of either Skypersonic or any Subsidiary are known to be engaged in organizing any labor union or other employee group that is seeking recognition as a bargaining

unit. Without limiting the generality of Section 5.11, (a) no unfair labor practice complaints are pending or, to Principal Sellers' Knowledge, threatened against either Skypersonic or any Subsidiary, and (b) no Person has made any claim and, to Principal Sellers' Knowledge, there is no basis for any claim, against either Skypersonic or any Subsidiary under any statute, regulation or ordinance relating to employees or employment practices, including without limitation those relating to age, sex and racial discrimination, conditions of employment, and wages and hours.

5.19 Contracts.

(a) <u>Schedule 5.19</u> sets forth a list of all material written agreements, arrangements or commitments to which Skypersonic or any Subsidiary is a party or by which any of its assets is bound or affected (all such contracts, agreements, arrangements or commitments as are required to be set forth on <u>Schedule 5.19</u> being referred to herein collectively as the "<u>Contracts</u>"), including, without limitation with respect to Skypersonic and each Subsidiary:

(b) each partnership, joint venture or similar agreement of Skypersonic and each Subsidiary with another Person;

(c) each contract or agreement under which Skypersonic or any Subsidiary has created, incurred, assumed or guaranteed (or may create, incur, assume or guarantee) indebtedness of more than US\$25,000 in principal amount or under which Skypersonic or any Subsidiary has imposed (or may impose) a Lien on any of its assets, whether tangible or intangible securing indebtedness in excess of US\$25,000;

(d) each contract or agreement which involves an aggregate payment or commitment per contract or agreement on the part of Skypersonic and each Subsidiary of more than US\$25,000 per year;

(e) each contract or agreement which involves or contributes to Skypersonic or any Subsidiary, aggregate annual remuneration which exceeds 5% of such Company's consolidated annual net revenues for the twelve months ended December 31, 2018 and 2019;

(f) all leases and subleases from any third person to Skypersonic or any Subsidiary, in each case requiring annual lease payments in excess of US\$25,000;

(g) each contract or agreement to which such Skypersonic or any Subsidiary or any of its Affiliates is a party limiting the right of Skypersonic or any Subsidiary (i) to engage in, or to compete with any person in, any business, including each contract or agreement containing exclusivity provisions restricting the geographical area in which, or the method by which, any business may be conducted by such Company or (ii) to solicit any customer or client;

(h) all collective bargaining or other labor union contracts or agreements to which such Company is a party or applicable to persons employed by Skypersonic or such Subsidiary;

(i) all licenses, licensing agreements and other agreements providing in whole or part for the use of any Intellectual Property of Skypersonic or such Subsidiary; and

(j) all other contracts or agreements which individually or in the aggregate are material to Skypersonic or such Subsidiary or the conduct of its business, other than those which are terminable upon no more than 30 days' notice by such Company without penalty or other adverse consequence.

<u>Schedule 5.19</u> further identifies each of the Contracts which contain anti-assignment, change of control or notice of assignment provisions. The Contracts are each in full force and effect and are the valid and legally binding obligations of Skypersonic and each Subsidiary which is a party thereto. To the Knowledge of Principal Sellers and the Company, neither Skypersonic nor any Subsidiary is a party to, nor is its business or any of its assets bound by, any oral agreement. Neither Skypersonic nor any Subsidiary is in default under any Contract to which it is a party, and no event has occurred which with the giving of notice or lapse of time or both would constitute such a default.

5.20 Tax Matters. Except as set forth on Schedule 5.20:

(a) All Tax Returns required to be filed on or before the Closing Date by or with respect to Skypersonic and any Subsidiary of Skypersonic have (or by the Closing Date will have) been duly filed or the time for filing such Tax Returns shall have been validly extended to a date after the Closing Date. Except for Taxes reflected or reserved against in the Financial Statements, the Companies and their Subsidiaries have paid all Taxes due and with respect to Taxes not yet due, all such Taxes not yet paid have been appropriately reserved against in accordance with GAAP. Neither the Company nor any Subsidiary of the Company is subject to any joint venture, partnership, or other arrangement or Contract which is treated as a partnership for federal income tax purposes. Neither the Company nor any Subsidiary of the Company is a party to any tax sharing agreement.

(b) As of the date hereof, there is no agreement or other document extending, or having the effect of extending, the period of assessment or collection of any Taxes of the Company or any Subsidiary of the Company, and no power of attorney with respect to any such Taxes, has been executed or filed with the IRS or any other taxing authority that remains in force.

(c) During the last 5 years, there has been no and as of the Effective Date there are no presently pending audits or other administrative proceedings or court proceedings with respect to any Taxes of the Company or any Subsidiary of the Company. No claim has ever been made by a governmental entity in a jurisdiction where a Company does not file a Tax Return that any of the Companies is or may be subject to taxation by that jurisdiction.

(d) All Taxes required to be withheld by the Company or any Subsidiary of the Company have been duly and timely withheld, and such

withheld Taxes have been duly and timely paid to the appropriate governmental entity.

(e) There are no Tax liens upon the assets of the Company or any Subsidiary of the Company except liens for Taxes not yet due and payable.

5.21 Guaranties. Neither the Company nor any Subsidiary of the Company has executed guaranteed the obligations of a third party.

5.22 Insurance. Set forth on Schedule 5.22 is a list of all insurance policies providing insurance coverage of any nature to the Company. The Company has previously delivered to Parent a true and complete copy of all of such insurance policies as amended. Such policies are sufficient for the compliance by the Company with all requirements of Law and all Contracts. All of such policies are in full force and effect and, to Principal Sellers' Knowledge, are valid and enforceable in accordance with their terms, and the Company has complied with all terms and conditions of such policies, including the payment of premium payments. None of the insurance carriers has indicated to the Company an intention to cancel any such policy. The Company has no claim pending or anticipated against any of the insurance carriers under any of such policies and, to Principal Sellers' Knowledge, there has been no actual or alleged occurrence of any kind which may give rise to any such claim.

5.23 <u>Inventories.</u> <u>Schedule 5.23</u> sets forth a list of all of the Company's and Subsidiary's inventory as of the Effective Date. The inventories of the Company and each Subsidiary of the Company as of the Closing Date shown on the Closing Trial Balance will be stated at the lower of cost (on a first-in first-out basis) or market. The inventories of the Company and each Subsidiary shown on the balance sheets included in the Financial Statements are fit for their particular use, do not and will not include any items below standard quality, defective, damaged or spoiled, obsolete or of a quality or quantity not usable or salable in the ordinary course of the business of the Company or Subsidiary as currently conducted or any items whose expiration date has passed or will pass within twelve months of the date hereof and of Closing (which, with respect to items which do not have an expiration date, shall in any event not include quantities of items not usable or salable within twelve months from the date hereof), the value of which has not been fully written down or reserved against in the Financial Statements. The Company and each Subsidiary has and will continue to have adequate quantities and types of inventory to enable it to conduct its business consistent with past practices and anticipated operations.

5.24 Regulatory Compliance.

(a) The Company and each Subsidiary is in compliance with all applicable Laws, rules, regulations, and policies administered or enforced by the U.S. FAA any other U.S. federal governmental entity that regulates the development of UAV products in any jurisdiction and with applicable federal Laws relating to anti-kickback sales and marketing practices, insurance and bonding, advertising and promotion, preand post-marketing reporting, and all other pre- and post-marketing reporting requirements, as applicable.

(b) <u>Schedule 5.24(b)</u> lists each product developed, manufactured, licensed, distributed or sold by the Company and each Subsidiary (collectively, the "<u>Products</u>"). Each Product manufactured by or on behalf of the Company or Subsidiary has been manufactured in accordance with (i) the product registration applicable to such Product, (ii) the applicable provisions of current "CE" good manufacturing practices or other governmental authority for products sold in Europe, and (iii) without limiting the generality of Section 5.11, the provisions of all applicable Laws.

(c) The Company and each Subsidiary has obtained all Regulatory Approvals required by the FAA to conduct the business as it is currently conducted. All of the Regulatory Approvals have been duly and validly issued and are in full force and effect, and the Company and each Subsidiary is in compliance with each such permit held by or issued to it. Except as listed on <u>Schedule 5.24(c)</u>, the Company and each Subsidiary is the sole and exclusive owner of the Regulatory Approvals and holds all right, title and interest in and to all such Regulatory Approvals. Neither the Company nor any Subsidiary of the Company has granted any third party any right or license to use, access or reference any of the Regulatory Approvals, including without limitation, any of the know-how contained in any of the Regulatory Approvals or rights (including any regulatory exclusivities) associated with each such Regulatory Approvals.

(d) There is no action or proceeding by any governmental or regulatory authority pending or, to the Knowledge of the Company, Principal Sellers, threatened seeking the recall of any of the Products or the revocation or suspension of any Regulatory Approval. The Company has made available to Parent complete and correct copies of all Regulatory Approvals. In addition, (i) the Company or any Subsidiary has made available to Parent a complete and correct copy of the Product Data; (ii) to the Knowledge of the Company and the Principal Sellers, all Laws applicable to the preparation and submission of the Regulatory Approvals to the relevant regulatory authorities have been complied with; (iii) to the Knowledge of the Company and the Principal Sellers, the Company or any Subsidiary has filed with the relevant regulatory authorities all required notices, supplemental applications, and annual or other reports, including adverse experience reports, with respect to the Regulatory Approvals.

(e) To Principal Sellers' Knowledge, there exist no set of facts: (i) which could furnish a basis for the recall, withdrawal or suspension of any product registration, product license, manufacturing license, wholesale dealers license, export license or other license, approval or consent of any domestic or foreign governmental or regulatory authority with respect to either Company or any of the Products; or (ii) which could furnish a basis for the recall, withdrawal or suspension of any Product from the market, the termination or suspension of any testing of any Product, or the change in marketing classification of any Product.

(f) Except as disclosed in <u>Schedule 5.24(f)</u> hereto, neither Company or any Subsidiary has received any claims based on alleged failure to meet the specifications or breach of product warranty arising from any applicable manufacture or sale of their Products.

(g) As of the Closing Date, all Product Inventory will conform to the specifications therefor contained in the Regulatory Approvals and to the Regulatory Approvals and with the requirements of all applicable governmental or regulatory authorities.

(h) The Company and each Subsidiary is and has been in compliance with all Laws requiring the maintenance or submission of reports or records under requirements administered by the FAA or any other applicable U.S. federal governmental authority. Neither Company nor any Subsidiary nor, to Principal Sellers' Knowledge, any of their respective employees or agents, have made an untrue or fraudulent statement to the FAA or any other applicable governmental authorities, or in any records and documentation prepared or maintained to comply with the applicable Laws, or failed to disclose a fact required to be disclosed to the FAA or any other similar governmental authorities.

(i)Neither Company nor any Subsidiary of the Company has been convicted of any crime or engaged in any conduct that could result or resulted in debarment, exclusion or disqualification by the FAA or any other governmental authority and there are no proceedings pending or, to the Knowledge of either Company or the Principal Sellers, threatened against the Company or any Subsidiary of the Company that reasonably might be expected to result in criminal or civil liability or debarment, exclusion or disqualification by the FAA or any other governmental authority. Neither Company nor any Subsidiary of the Company has received written notice of or been subject to any other enforcement action involving the FAA or any other governmental authorities, including any suspension, consent decree, notice of criminal investigation, indictment, sentencing memorandum, plea agreement, court order or target or no-target letter, and none of the foregoing are pending or, to any Company's or Principal Sellers' Knowledge, threatened in writing against any Company.

(j) The Company and each Subsidiary of the Company has security measures and safeguards in place which are intended to protect Personally Identifiable Information it collects from customers and other parties from illegal or unauthorized access or use by its personnel or third parties in a manner that violates the privacy rights of third parties. To the knowledge of Company and the Principal Sellers, the Company and each Subsidiary of the Company has complied in all material respects with all Applicable Laws relating to privacy and consumer protection and neither has collected, received, stored, disclosed, transferred, used, misused or permitted unauthorized access to any information protected by Applicable Laws related to privacy, whether collected directly or from third parties, in an unlawful manner. The Company and each Subsidiary of the Company has taken all reasonable steps to protect Personally Identifiable Information against loss or theft and against unauthorized access, copying, use, modification, disclosure or other misuse.

(k) Neither Company nor any Subsidiary of the Company is party to any corporate integrity agreements, monitoring agreements, consent decrees, settlement orders or similar agreements with or imposed by any governmental authority.

5.25 Intellectual Property Rights.

(a) <u>Schedule 5.25</u> sets forth a complete and correct list of all Intellectual Property that is owned by the Company or any Subsidiary of the Company and the Intellectual Property that the Company or any Subsidiary of the Company has a license, sublicense or other permission to use. Except as set forth in <u>Schedule 5.25</u>, the Company or any Subsidiary of the Company owns all right, title and interest in and to, or has a license, sublicense or other permission to use, all of the Intellectual Property, free and clear of all Liens or other encumbrances. All necessary registration, maintenance and renewal fees in connection with such Intellectual Property have been paid and all necessary documents and certificates in connection with such Intellectual Property.

(b) The Intellectual Property constitutes all patents and patent applications, and technology, know-how and information owned or licensed to the Company or any Subsidiary of the Company relating to the manufacture, use or sale of the Products. There have been no claims made against either Company or any Subsidiary of the Company asserting the invalidity, abuse, misuse, or unenforceability of any of the Intellectual Property, and, to the Knowledge of the Company and the Principal Sellers, no grounds for any such claims exist. Neither Company any Subsidiary of the Company has made any claim of any violation or infringement by others of its rights in the Intellectual Property, and, to the Knowledge of the Company and Principal, no grounds for any such claims exist. Neither the Company nor any Subsidiary of the Company has received any notice that it is in conflict with or infringing upon the asserted rights of others in connection with the Intellectual Property and, to the Knowledge of the Company and Principal Sellers, the use of the Intellectual Property by the Company or any Subsidiary of the Company is not infringing and has not infringed upon any rights of any other Person. No interest in any of the Intellectual Property has been assigned, transferred, licensed or sublicensed by either Company or any Subsidiary of the Company to any Person. Neither Company nor any Subsidiary of the Company nor Principal Sellers has Knowledge of any act or failure to act by any of them or any of their respective directors, officers, employees, attorneys or agents during the prosecution or registration of, or any other proceeding relating to, any of the Intellectual Property or of any other fact which could render invalid or unenforceable, or negate the right to issuance of any of, any of the Intellectual Property.

(c) The Company and each Subsidiary of the Company has taken reasonable steps to protect the confidentiality and value of the Company's material trade secrets and any other material confidential information that are owned, used, or held by such Company in confidence,. To the Knowledge of the Company and Principal Sellers, such trade secrets have not been disclosed to, or discovered by, any Person who is not affiliated with the Company except pursuant to a non-disclosure, license, or any other Contract which has not been breached.

5.26 <u>Power of Attorney</u>. The Company has not issued, granted or executed any powers of attorney on behalf of such Company which is in force at the Closing Date, other than in connection with routine legal filings.

5.27 <u>Absence of Material Adverse Effects</u>. Since December 31, 2020, the Company or any Subsidiary has conducted its business only in the ordinary and usual course and in a manner consistent with past practices.

5.28 Accounts and Notes Receivable and Payable. Set forth on Schedule 5.28 is a true and complete aged list of unpaid accounts and notes receivable owing to and owed by the Company or any Subsidiary as of the date hereof. All of such accounts and notes receivable and payable constitute bona fide, valid and binding claims arising in the ordinary course of such Company's business. Except as set forth on Schedule 5.28, there is no agreement for deduction, free goods, discounts, or other deferred price or adjustment to such receivables.

5.29 Banks. Schedule 5.29 sets forth (i) the name of each bank, trust corporation or other financial institution and stock or other broker with which the Company or any Subsidiary of the Company has an account, credit line or safe deposit box or vault, (ii) the names of all Persons authorized to draw thereon or to have access to any safe deposit box or vault, (iii) the purpose of each such account, safe deposit box or vault, and (iv) the names of all Persons authorized by proxies, powers of attorney or other like instrument to act on behalf of the Company or any Subsidiary in matters concerning their business or affairs. Except as otherwise set forth in <u>Schedule 5.31</u>, no such proxies, powers of attorney or other like instruments are irrevocable.

5.30 Accuracy of Information Furnished. To the knowledge of Company and Principal Sellers, no representation, statement or information by the Company contained in this Agreement (including the various exhibits attached hereto) or any agreement executed by the Company in connection herewith or in any certificate or other document delivered by the Company pursuant hereto or thereto or made or furnished to Parent or its representatives by the Company, contains or shall contain any untrue statement of a material fact or omits or shall omit any material fact necessary to make the information contained herein and therein not misleading. Copies of all documents of the Company listed or described in the various exhibits attached hereto and provided by the Company to Parent are true, accurate and complete.

ARTICLE 6

ADDITIONAL AGREEMENTS

6.1 Noncompetition.

(a) The Sellers acknowledges that in order to assure Buyer that Buyer will retain the value of the Company or any Subsidiary as a "going concern," Santangelo agrees not to utilize his special knowledge of the business of any Company and his relationships with customers, suppliers and others to compete with either Parent, and of its Subsidiaries, or the Company in a Prohibited Business. For a period of three (3) years beginning on the Closing Date (the "<u>Noncompete Period</u>"), neither Santangelo nor any of Santangelo's Affiliates shall engage or have an interest in (directly or indirectly) any entity that is engaged in a Prohibited Business anywhere in the United States or any other geographic area where Parent, any of its Subsidiaries, or any Company does business, alone or in association with others, as principal, officer, agent, employee, director, partner or shareholder, or through the investment of capital, lending of money or property, rendering of services or otherwise. "Prohibited Business" means engaging in any activities in providing products, services and solutions to the drone industry, first person view ("<u>FPV</u>") business relating to UAV, navigation and software solutions that provide analytics, storage or services for or in conjunction with the drone industry in the geographical area in which Parent, the Company and any Subsidiary of the Company conducts such business as of the Closing Date. Santangelo acknowledges and agrees that the limitations imposed by this Section 6.1 as to time, geographical area, and scope of activity being restrained are reasonable and do not impose a greater restraint than is necessary to protect the goodwill or other business interests of Parent, any Subsidiary and the Companies.

(b) During the Noncompete Period, except in connection with services to a Company, Santangelo shall not directly and shall not, knowingly or intentionally, permit any of his directors, officers, employees, agents or others under his control to, directly or indirectly, (i) call upon, accept business from, or solicit for a Prohibited Business the business of any Person who is, or who had been at any time during the preceding two (2) years, a customer of any Company, or otherwise divert or attempt to divert sales from the Company; or (ii) recruit or otherwise solicit or induce any person who is an employee of, independent contractor or otherwise engaged by, the Company to terminate his or her employment or other relationship with the Company, or hire any person who has left the employ of the Company during the preceding two (2) years. Except in connection with services to the Company, Sellers shall not at any time, directly or indirectly, use or purport to authorize any Person to use any name, mark, logo, trade dress or other identifying words or images which are the same as or similar to those used currently or in the past by Parent or the Company in connection with any product or service, whether or not such use would be in a business competitive with that of Parent or the Company.

6.2 <u>Confidentiality</u>. Sellers acknowledge that all confidential or proprietary information with respect to the business and operations of the Company or any Subsidiary of the Company is valuable, special and unique. Sellers shall not, at any time before or after the Closing Date (except as required in connection with his services to either Company) disclose, directly or indirectly, to any Person, or use or purport to authorize any Person to use any confidential or proprietary information with respect to the Company, Parent and of Parent's Subsidiaries, whether or not for Sellers' own benefit, without the prior written consent of Parent, including without limitation, information as to the financial condition, results of operations, customers, suppliers, products, products under development, inventions, sources, leads or methods of obtaining new products or business, pricing methods or formulas, cost of supplies, marketing strategies or any other information which is or shall become generally available to the public other than as a result of an unauthorized disclosure by Sellers or a Person to whom Sellers has provided such information. Sellers acknowledge that Parent and Parent would not enter into this Agreement without the assurance that all such confidential and proprietary information will be used for the exclusive benefit of Parent, Parent and the Companies.

6.3 <u>Continuing Obligations</u>. The restrictions set forth in Sections 6.1 and 6.2 are considered by the Parties to be reasonable for the purposes of protecting the value of the business and goodwill of the Company or any Subsidiary and Parent. Sellers acknowledge that Parent would be irreparably harmed and that monetary damages would not provide an adequate remedy in the event the covenants contained in Sections 6.1 and 6.2 were not complied with in accordance with their terms. Accordingly, Sellers agree that any breach or threatened breach of any provision of Sections 6.1 and 6.2 shall entitle Parent to injunctive and other equitable relief, without requirement of posting a bond or other surety, to secure the enforcement of these provisions, in addition to any other remedies which may be available to Parent. In the event of litigation or other proceeding concerning Section 6.1 or 6.2, the prevailing party therein shall be entitled to recover the reasonable attorneys' fees and expenses incurred therein from the party who did not so prevail. In addition to its other rights and remedies, Parent shall have the right to require any party who breaches any of the covenants contained in Sections 6.1 and 6.2 to account for and pay over to Parent all compensation, profits, money, accruals and other benefits derived or received, directly or indirectly, by such breaching party from the action constituting such breach. If a party breaches the covenants set forth in Section 6.1, the running of the Noncompete Period shall be tolled with

respect to such party for so long as such breach continues. It is the desire and intent of the parties that the provisions of Sections 6.1, 6.2 and 6.3 be enforced to the fullest extent permissible under the laws and public policies of each jurisdiction in which enforcement is sought. If any provisions of Sections 6.1, 6.2 and 6.3 relating to the time period, scope of activities or geographic area of restrictions is declared by a court of competent jurisdiction to exceed the maximum permissible time period, scope of activities or geographic area, the maximum time period, scope of activities or geographic area, the maximum time period, scope of activities or geographic area, the maximum time period, scope of activities or geographic area, as the case may be, shall be reduced to the maximum which such court deems enforceable. If any provisions of Sections 6.1, 6.2 and 6.3 other than those described in the preceding sentence are adjudicated to be invalid or unenforceable, the invalid or unenforceable provisions shall be deemed amended (with respect only to the jurisdiction in which such adjudication is made) in such manner as to render them enforceable and to effectuate as nearly as possible the original intentions and agreement of the parties.

6.4 Investment Intent; Accredited Investor Status; Restrictions on Sale; Registration of Securities.

(a) Each of the Sellers represents that it has such knowledge and experience in business or financial matters that it is capable of evaluating the merits and risks of an investment in the Stock Consideration. Each of the Sellers understands and acknowledges that the Stock Consideration has not been registered with the SEC and that Sellers may not sell, transfer or otherwise dispose of all or any portion of the Stock Consideration except (i) in accordance with the provisions of Rule 144 under the Securities Act, (ii) pursuant to an effective registration statement under the Securities Act or (iii) upon receipt by Parent of an opinion of counsel acceptable to Parent to the effect that such sale, transfer or disposition is otherwise exempt from registration under the Securities Act. Certificates representing the Stock Consideration shall bear a restrictive legend.

(b) If an underwriter engaged by Parent in connection with any underwritten or best efforts registered offering by the Parent of its equity securities pursuant to an effective registration statement filed under the Securities Act requires a shareholder of Red Cat in Santangelo's position to become subject to a market stand-off agreement, Santangelo shall execute and deliver a written agreement by which he will agree that he will not, during a period not to exceed ninety (90) days after the effective date of the registration statement (or such other period as may be necessary to accommodate regulatory restrictions on (i) the publication or other distribution of research reports and (ii) analyst recommendations and opinions, including, but not limited to, the restrictions contained in NASD Rule 2711(f)(4) or NYSE Rule 472(f)(4), or any successor provisions or amendments thereto), sell, make any short sale of, loan, hypothecate, pledge, grant any option for the purchase of, or otherwise dispose or transfer for value or otherwise agree to engage in any of the foregoing transactions with respect to, any Stock Consideration without the prior written consent of Parent or its underwriters. Notwithstanding any provision hereof to the contrary, Santangelo shall only be required to enter into any such market stand-off agreement if all officers and directors of Parent and each of the Parent's direct and indirect subsidiaries and holders of at least one percent (1%) of the Company's voting securities are also required by such underwriter to enter into similar agreements.

6.5 Tax Matters.

(a) Intentionally omitted.

(b) In the case of any taxable period that includes (but does not end on) the Closing Date (a "<u>Straddle Period</u>"), the amount of any Taxes based on or measured by income or receipts of the Company for the portion of the Straddle Period ending on the Closing Date shall be determined based on an interim closing of the books as of the close of business on the Closing Date (and for such purpose, the taxable period of any partnership or other pass-through entity in which the Company holds a beneficial interest shall be deemed to terminate at such time) and the amount of other Taxes of the Company for a Straddle Period that relates to the portion of the Straddle Period ending on the Closing Date shall be deemed to be the amount of such Tax for the entire taxable period multiplied by a fraction the numerator of which is the number of days in the taxable period ending on the Closing Date and the denominator of which is the number of days in such Straddle Period.

(c) From the date of this Agreement through and after the Closing Date, the Sellers shall cause the Company to prepare and file as required by applicable Law with the appropriate taxing authority (or cause to be prepared and filed) in a timely manner (i) all consolidated, combined, unitary, affiliated or similar Tax Returns that include the Company or any Subsidiary, on the one hand, and any Sellers, on the other hand, for all periods ending on or prior to the Closing Date and (ii) all other Tax Returns of the Company and its Subsidiary that are required to be filed on or prior to the Closing Date and will provide Parent such Tax Returns no later than 30 days prior to the due date for filing such Tax Returns to provide Parent with a meaningful opportunity to analyze and comment on such Tax Returns and for such Tax Returns to be modified, as appropriate, before filing. Sellers will consider any comments of Parent relating to Tax Returns described in the preceding sentence.

(d) Parent shall prepare and file as required by applicable Law with the appropriate taxing authority (or cause to be prepared and filed) in a timely manner all Tax Returns of the Company and its Subsidiary, other than those described in <u>Section 6.6(c)</u>. Parent shall prepare any Tax Returns relating to Taxes for which the Sellers may have an indemnification obligation to Parent under this Agreement consistent with the Sellers' recent past practices, except as otherwise required by applicable Law, and make any Tax Returns relating to Taxes for which the Sellers may have an indemnification obligation under this Agreement available for review by the Sellers no later than 30 days prior to the due date for filing such Tax Returns to provide the Sellers with a meaningful opportunity to analyze and comment on such Tax Returns and for such Tax Returns to be modified, as appropriate, before filing. Parent will consider any comments of the Sellers relating to Tax Returns described in the preceding sentence.

(e) After the Closing, Parent and its Affiliates shall not amend any Tax Return of the Company or its Subsidiary that could impact any Sellers, including any indemnification obligation of the Sellers pursuant to this Agreement, without the Sellers' consent, not to be unreasonably withheld; provided however that Parent and its Affiliates may file any such amendment that is required by Law or regulation to be filed.

(f) Parent and the Sellers shall (and shall cause their respective Affiliates to) (i) provide the other party and its Affiliates with such assistance as may be reasonably requested in connection with the preparation of any Tax Return or any audit or other examination by any

taxing authority or any judicial or administrative proceeding relating to Taxes, which shall include Parent's obligation to provide the Sellers, consistent with past practice (including with respect to timing for delivery), the information needed for the preparation of all consolidated, combined, unitary, affiliated or similar Tax Returns that include the Company or its Subsidiary, on the one hand, and any Sellers or any Sellers Affiliate, on the other hand, through the Closing Date and (ii) retain (and provide the other party and its Affiliates with reasonable access to) all records or information which may be relevant to such Tax Return, audit, examination or proceeding, provided that the foregoing shall be done in a manner so as not to interfere unreasonably with the conduct of the business of the parties. The Sellers shall not destroy any records or information relating to any Tax compliance matter or Tax liability of the Company or its Subsidiary without first notifying Parent and offering Parent the opportunity to make copies or to take possession of such records or information. Notwithstanding the above, the Sellers shall have no obligation to share with Parent any consolidated, combined or unitary return or associated forms, documents or workpapers, that include the Sellers or any Sellers Affiliate.

(g) All refunds of Taxes (including interest actually received thereon from a relevant taxing authority) paid prior to the Closing, or for which the Sellers have provided indemnification pursuant to this Agreement shall be for the account of the Sellers, and Parent shall pay such amounts, less Parent's reasonable out-of-pocket expenses, including professional fees, incurred in connection with obtaining any such refund and less any Taxes incurred by Parent, its Affiliates, the Company or its Subsidiary in connection with the receipt of any such refund or interest, to the Sellers if such refunds are received by Parent, the Company or its Subsidiary.

(h) All tax sharing agreements or similar agreements with respect to or involving the Company or its Subsidiary shall be terminated as of the Closing Date and, after the Closing Date, the Company and its Subsidiary shall not be bound thereby or have any liability thereunder.

(i) The Sellers shall be liable for all transfer, documentary, sales, use, stamp, registration, value added and other similar Taxes and fees (including any penalties and interest) incurred in connection with sale of the Purchased Shares contemplated by this Agreement ("<u>Transfer</u> <u>Taxes</u>") and Sellers shall, at their own expense, file all necessary Tax Returns and other documentation with respect to all such Taxes, fees and charges, and, if required by applicable law, Parent will, and will cause its affiliates to, join in the execution of any such Tax Returns and other documentation.

6.6 Intentionally omitted.

6.7 <u>Drag Along Rights</u>. If Stockholders of Parent (the "<u>Selling Stockholders</u>") holding a Stockholder Majority (as defined below) propose to sell all their shares of Parent Common Stock to any unaffiliated or unrelated person in an arm's length transaction, then the Selling Stockholders may elect to require (a "<u>Drag-Along Right</u>") that Sellers sell all of the shares received as Stock Consideration in such sale transaction(s) for the same price per share as the Selling Stockholders (a "<u>Drag-Along Transaction</u>"). If the Selling Stockholders desire to exercise such Drag-Along Right, they shall give written notice to the Sellers of the proposed Drag-Along Transaction giving rise to the Drag-Along Right at least twenty (20) days prior to the consummation thereof (a "<u>Drag-Along Notice</u>"). The Drag-Along Notice shall set forth the principal terms of such proposed transaction including the amount of Parent Common Stock to be sold by the Selling Stockholders, the price per shares to be paid, the name and address of the prospective Parent and any other material terms and conditions (in reasonable detail) pertaining to such proposed transaction to which reference is made in the Drag-Along Notice, the Sellers shall be bound and obligated to sell all of his Parent Common Stock in the proposed transaction on the same terms and conditions as the Selling Stockholders. If properly exercised, the Sellers are hereby deemed to have consented to the transaction. A "Stockholder Majority" shall mean holders of fifty (50%) percent or greater of the outstanding shares of Parent Common Stock.

6.8 <u>Tag Along</u>. If Selling Stockholders holding a Stockholder Majority propose to sell a number of shares equal to a Stockholder Majority to any unaffiliated or unrelated person in an arm's length transaction (a "<u>Tag-Along Transaction</u>"), then the Selling Stockholders must give written notice to the Sellers of the proposed Tag-Along Transaction at least twenty (20) days prior to the consummation thereof (a "<u>Tag-Along Notice</u>"). The Tag-Along Notice shall set forth the principal terms of such proposed transaction including the amount of Parent Common Stock to be sold by the Selling Stockholders, the price per shares to be paid, the name and address of the prospective purchaser and any other material terms and conditions (in reasonable detail) pertaining to such proposed transfer, including without limitation, the proposed closing date (the "<u>Tag-Along Terms</u>"). Within twenty (20) days after receipt of the Tag-Along Notice, Sellers may elect by written notice to the Selling Stockholders, to require that the Selling Stockholders include a pro rata portion of Sellers' Parent Common Stock in the Tag-Along Transaction (the "<u>Tag-Along Right</u>"). The purchaser shall have the right to either increase the total number of shares of Parent Common Stock being purchased or reduce the numbers of share being purchased from the Selling Stockholders by an amount necessary to include Sellers' shares in the Tag-Along Transaction. If Sellers exercises the Tag-Along Right, the Selling Stockholders may not consummate the proposed transaction to which reference is made in the Tag-Along Notice without including the Sellers.

ARTICLE 7

SURVIVAL; INDEMNIFICATION

7.1 Intentionally omitted.

7.2 <u>Survival of the Representations and Warranties</u>. The representations and warranties and indemnification obligations of the Sellers, the Companies and Parent shall survive the Closing Date for a period of one (1) year from the Closing Date; provided, however, that (i) the representations in Section 4.1 (Authorization: Enforceability), Section 4.2 (No Consent, Violation or Conflict), Section 4.5 (Capitalization), Section 4.6 (Rights, Warrants, Options), Section 5.1 (Organization), Section 5.2 (Authorization; Enforceability); Section 5.3 (No Consent, Violation or Conflict), Section 5.7 (Capitalization) and 5.8 (Rights, Warrants, Options) (collectively, the "Fundamental Representations") shall survive indefinitely and (ii) the representations and warranties set forth in Sections 5.16 (Compliance with Environmental Laws), 5.17 (Employment Matters) and 5.20 (Tax Matters) shall survive the Closing Date until the expiration of the period

specified in the applicable statute of limitations.

7.3 <u>General Release</u>. Other than those obligations of the Company specified on <u>Schedule 7.3</u>, Sellers hereby unconditionally and irrevocably release and forever discharge, effective as of the Closing Date, the Company and any Subsidiary of the Company and its officers, directors, employees and agents, from any and all rights, claims, demands, judgments, obligations, liabilities and damages, whether accrued or unaccrued, asserted or unasserted, and whether known or unknown, relating to the Company which ever existed, now exist, or may hereafter exist, by reason of any tort, breach of contract, violation of law or other act or failure to act which shall have occurred, or which shall be based on any fact, occurrence, act or omission occurring or arising, at or prior to the Closing Date. None of the information supplied by the Company and any Subsidiary or its professional advisors to Sellers or its agents, representatives or advisors in connection with the representations and warranties set forth in Article 5 or otherwise in relation to the business or affairs of such Company shall be deemed a representation, warranty or guarantee of its accuracy by such Company to such Sellers, and such Sellers waives any claims against such Company which it might otherwise have in respect of it.

7.4 Indemnification.

(a) Indemnification by the Principal Sellers. Subject to the other terms and conditions specified herein, the Principal Sellers shall, in the proportion which the number of Purchased Shares held by each Principal Seller bears to the number of Purchased Shares held by all Principal Sellers indemnify and hold harmless Parent and Buyer from, against and in respect of, the full amount of:

(i) (A) any and all actions, suits, proceedings, demands, liabilities, damages, claims, deficiencies, fines, penalties, interest, assessments, judgments, losses, Taxes, costs and expenses, including, without limitation, reasonable fees and disbursements of counsel (collectively, the "Indemnified Losses") incurred by Parent or Buyer arising from or in connection with any breach or violation of any of the representations and warranties of the Company or Principal Sellers contained in this Agreement or (B) any and all Indemnified Losses arising from or in connection with any breach or violation of the covenants or agreements of the Company or Principal Sellers contained in this Agreement and, if reasonably capable of cure, the failure of the Company or Principal Sellers to cure the same within thirty (30) days after written notice thereof is provided by Parent or Buyer;

(ii) any and all Indemnified Losses for Taxes attributable to all Tax years or portions thereof ending on or prior to the Closing Date imposed on the Company which are not reflected in the Financial Statements, Closing Trial Balance or the Disclosure Schedules;

(iii) any and all capital or other Taxes related to or arising from the sale and transfer of the Purchased Shares contemplated hereby by reason of any Liability of the Company or its shareholders for such Taxes as assessed by any taxing authority against Sellers and/or the Company either before or after the Closing Date;

(iv) any and all Indemnified Losses which relate to any legal and/or governmental proceedings of the nature described in Section 5.12 which are not set forth on Schedule 5.12, existing on or prior to the Closing Date, and/or which are brought after the Closing Date for acts and omissions of the Company which occurred prior to the Closing Date; and

(v) any and all Indemnified Losses arising from Liabilities of the Company not disclosed to Parent, included in the Disclosure Schedules or reserved for on the Closing Trial Balance, related to the business or operations of any Company prior to the Closing Date.

(b) Indemnification by Sellers. Subject to the other terms and conditions specified herein, each Seller shall indemnify and hold harmless Parent and Buyer from, against and for the full amount of any and all Indemnifiable Losses incurred by Parent or Buyer arising from or in connection with:

(i) any breach or violation of any of the representations and warranties of such Sellers contained in this Agreement; or

(ii) any breach or violation of the covenants or agreements of such Seller contained in this Agreement and, if reasonably capable of cure, the failure of such Seller to cure the same within thirty (30) days after written notice thereof is provided by Parent or Buyer.

(c) Indemnification by Parent. Parent and Buyer agree to defend, indemnify and hold harmless Sellers (and prior to Closing, Parent and Buyer shall also indemnity the Company) from, against and in respect of, the full amount of

(i) any and all Indemnified Losses arising from or in connection with any breach or violation of any of the representations or warranties of Parent or Buyer contained in this Agreement,

(ii) any and all Indemnified Losses arising from or in connection with any breach or violation of any of the covenants or agreements of Parent or Buyer contained in this Agreement and, if reasonably capable of cure, the failure of Parent or Buyer to cure the same within thirty (30) days after written notice thereof is provided by any Seller; and

(iii) any and all Indemnified Losses arising from or in connection with obligations under any Contract arising on or after the Closing Date; and

(iv) any and all Indemnified Losses arising from or in connection with the operation of the Company's business on or after the Closing Date.

(d) Indemnification Procedure. Any party seeking indemnification under this Agreement (the "Indemnified Party") will give prompt written notice to the party or parties against whom indemnity is sought (the "Indemnifying Party") of any Indemnified Losses which it discovers or of which it receives notice after the Closing, stating the nature, basis (including the section of this Agreement that has been or will

be breached, if any, and the facts giving rise to the claim that a breach has or will occur), and (to the extent known) amount thereof; *provided*, *however*, that no delay on the part of Indemnified Party in notifying any Indemnifying Party shall relieve the Indemnifying Party from any liability hereunder unless (and then solely to the extent) the Indemnifying Party is prejudiced by such delay. Notwithstanding any provision hereof to the contrary, no claim may be made or lawsuit or other proceeding initiated seeking indemnification under this Article 7 unless a written notice describing such claim in reasonable detail in light of the circumstances then known to the Indemnified Party, is provided to the Indemnifying Party prior to: (i) the expiration of the applicable survival period specified in Section 7.2 with respect to the representations and warranties contained herein; or (ii) the expiration of a period of eighteen (18) months after the Closing with respect to any other claim for indemnification.

(e) Indemnification Procedure as to Third Party Claims.

(i) Promptly after any Indemnified Party obtains knowledge of the commencement of any third party claim, action, suit or proceeding or of the occurrence of any event or the existence of any state of facts which may become the basis of a third party claim (any such claim, action, suit or proceeding or event or state of facts being hereinafter referred to in this Section 7.4 as a "Claim"), in respect of which an Indemnified Party is entitled to indemnification under this Agreement, such Indemnified Party shall promptly notify the Indemnifying Party of such Claim in writing; provided, however, that any failure to give notice (A) will not waive any rights of the Indemnified Party except to the extent that the rights of the Indemnifying Party are actually prejudiced thereby and (B) will not relieve the Indemnifying Party of its obligations as hereinafter provided in this Section 7.4 after such notice is given. With respect to any Claim as to which such notice is given by the Indemnified Party to the Indemnifying Party, the Indemnifying Party will, subject to the provisions of Section 7.4(e)(ii), assume the defense or otherwise settle such Claim with counsel reasonably satisfactory to the Indemnified Party (1) shall be permitted to join the defense and settlement of such Claim and to employ counsel reasonably satisfactory to the Indemnifying Party, and at the Indemnified Party's own expense, (2) shall cooperate fully with the Indemnifying Party in the defense and any settlement of such Claim in any manner reasonably requested by the Indemnifying Party; and (3) shall not compromise or settle any such Claim without the prior written approval of the Indemnifying Party;

(ii) If (A) the Indemnifying Party fails to assume the defense of such Claim or, having assumed the defense and settlement of such Claim, fails reasonably to contest such Claim in good faith, or (B) the remedy sought by the claimant with respect to such Claim is not solely for money damages, the Indemnified Party, without waiving its right to indemnification, may, but is not required to, assume the defense and settlement of such Claim, provided, however, that (1) the Indemnifying Party shall be permitted to join in the defense and settlement of such Claim in any manner reasonably requested by the Indemnified Party, and (3) the Indemnified Party shall not settle such Claim without the written consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed.

(iii) As used in this Section 7.4, the terms Indemnified Party and/or Indemnifying Party shall be deemed to include the plural thereof where the rights or obligations of more than one Indemnified Party and/or Indemnifying Party may be involved.

(f) <u>Tax-free Indemnification Payments</u>. All sums payable by an Indemnifying Party as indemnification under this Section 7.4 shall be paid free and clear of all deductions or withholdings (including any taxes or governmental charges of any nature) unless the deduction or withholding is required by law, in which event or in the event the Indemnified Party shall incur any liability for taxes chargeable or assessable in respect of any such payment, the Indemnifying Party shall pay such additional amounts as shall be required to cause the net amount received by the Indemnified Party to equal the full amount which would otherwise have been received by it had no such deduction or withholding been made or no such liability for taxes been incurred. This Section 7.4(f) shall not apply to any amount payable as indemnification if the payment is in replacement of an amount that would have been taxable to the Indemnified Party.

(g) <u>No Contribution</u>. The obligations of the Principal Sellers and Sellers to indemnify Parent and Buyer pursuant to the terms of this Agreement are primary obligations of the Sellers, subject to the limitations set forth herein. The Principal Sellers hereby waive any right to seek or obtain indemnification or contribution from the Company following the Closing for Indemnified Losses as a result of any breach by the Company of any representation, warranty or covenant contained in this Agreement.

7.5 Limitations on Liabilities.

(a) Notwithstanding anything to the contrary contained herein: (i) in no event shall the aggregate sums payable by the Principal Sellers under Section 7.4(a)(i)(A) or by the Sellers under Section 7.4(b)(i) (other than sums payable as a result of fraud or breaches of the Fundamental Representations enumerated in Section 7.2) exceed the aggregate amount of the Escrow Shares; and (ii) in no event shall the aggregate sum payable by the Principal Sellers under any other provision of Section 7.4(a) or by the Sellers under any other provision of Section 7.4(a) or by the Sellers under any other provision of Section 7.4(a) or by the Sellers under any other provision of Section 7.3(b) exceed the Stock Consideration at Closing.

(b) No claim for breach of representation or warranty shall be made by any Party if such Party had actual knowledge of the fact or event giving rise to such breach prior to or as of the Closing.

(c) Notwithstanding anything to the contrary contained herein, no party shall be obligated to indemnify and hold harmless any other Party under Section 7.4 for breaches of representations and warranties unless and until all Indemnified Losses in respect of which such party is obligated to provide indemnification exceed Twenty-Five Thousand Dollars (US\$25,000) (the "Basket Amount") following which (subject to the provisions of this Section 7.5) such party shall be obligated to indemnify and hold harmless, the other party for the amount of the Indemnified Losses in excess of the Basket Amount; provided however that the Basket Amount shall not apply to indemnity obligations for Indemnified Losses arising as a result of fraud or breaches of the representations and warranties in Sections 5.7 and 5.8.

(d) Notwithstanding anything to the contrary set forth herein, none of the limitations on indemnification set forth in Section 7.5(a) or (c) shall apply to matters relating to fraudulent misrepresentations.

(e) The amount of any payments required to be made under this Article 7 shall be reduced by the amount of any tax or insurance benefit actually received by (including by refund or by reduction of or offset against Taxes otherwise payable) the recipient by reason of the payment or incurrence by such recipient of the item for which the indemnity is being sought. Each party shall notify the other in writing of such receipt of any such tax or insurance benefits.

ARTICLE 8

INTERIM COVENANTS

8.1 Interim Operations of the Companies.

(a) The Company and Sellers covenant and agree that, from the date hereof until the Closing Date, the Company and any Subsidiary of the Company shall operate the business in accordance with its ordinary course and past practice. In addition, during the period commencing on the Effective Date and until the Closing Date, the Company and any Subsidiary of the Company shall, and Sellers shall cause such Company to, except to the extent Parent specifically gives its prior written consent to the contrary (provided Sellers acknowledge and agree that Parent's consent shall not be provided and payments not allowed under the provisions of (iv), (ix) and (xii) if there is an Estimated Working Capital Deficiency Amount and any such payments are not made in the ordinary course consistent with past practice or if such payments in the aggregate exceed any Estimated Working Capital Excess Amount):

(i) use its best efforts to preserve intact its business organization and the goodwill of its customers, suppliers and others having business relations with it;

(ii) use its best efforts to keep available to Buyer following the Closing the services of the Company's officers, employees, independent contractors and agents;

(iii) promptly furnish to Parent a copy of any correspondence received from or delivered to any governmental authority;

(iv) maintain and keep its properties and assets in substantially the same repair and condition as they were on the date of this Agreement (reasonable wear and tear from use in the ordinary course of business excepted);

(v) continue and maintain the approval process in the ordinary course of business with respect to the Products and any products being developed by such Company; and

(vi) continuously maintain insurance coverage substantially equivalent to the insurance coverage in existence on the Effective

Date.

(b) Additionally, during the period from the Effective Date to the Closing Date, except with the prior consent of Parent, the Company shall not, and Sellers shall not permit such Company to, directly or indirectly:

(i) amend or otherwise change such Company's Organizational Documents;

(ii) issue, sell or authorize for issuance or sale, shares of any class of its securities (including, but not limited to, by way of share split or dividend) or any subscriptions, options, warrants, rights or convertible securities, or enter into any agreements or commitments of any character obligating it to issue or sell any such securities;

(iii) redeem, purchase or otherwise acquire directly or indirectly any shares of its authorized share capital or any option, warrant or other right to purchase or acquire any such shares;

(iv) declare or pay any dividend or other distribution;

(v) sell, transfer, surrender, abandon or dispose of any of its assets or property rights (tangible or intangible), except for sales or dispositions of inventory in the ordinary course of business consistent with past practice;

(vi)grant, make or subject itself or any of its assets or properties to any Lien (other than Liens for Taxes not yet due);

(vii) create, incur or assume any liability or indebtedness which would remain with such Company after the Closing Date, except in the ordinary course of business consistent with past practice;

(viii) enter into, amend or terminate any Contract;

(ix) commit to make any capital expenditures in excess of US---\$10,000, which would be payable by the Company after the

Closing Date;

(x) grant any guaranty;

(xi) waive, release, assign, settle or compromise any material claim or litigation;

(xii) except as required by Law or the terms of any applicable Contract, increase the compensation payable or to become

payable to directors, officers, employees, consultants or agents or grant any rights to severance or termination pay to, or enter into any employment or severance agreement with any of the foregoing Persons or establish, adopt, enter into or amend any collective bargaining, bonus, profit sharing, thrift, compensation, share option, restricted share, pension, retirement, deferred compensation, employment, termination, severance or other plan, agreement, trust, fund, policy or arrangement for the benefit of any of the foregoing Persons;

(xiii) acquire (including, without limitation, by merger, consolidation or acquisition of stock or assets) any interest in any corporation, partnership, other business organization, Person or any division thereof or any assets;

(xiv) alter the manner of keeping its books, accounts or records, or change in any manner the accounting practices therein reflected;

(xv) make any Tax election or settle or compromise any material federal, state or local or federal income Tax Liability;

(xvi) change its accounting practices, methods or assumptions or write down any of its assets;

(xvii) enter into any commitment or transaction which would survive the Closing Date, except in the ordinary course of business consistent with past practice;

(xviii) grant any license or sublicense of any right under or with respect to any Intellectual Property or disclose any proprietary or confidential information to any third party;

(xix) take or omit to take any action which would render any of the Company's or any of any Sellers' representations or warranties untrue or misleading, or which would be a breach of any of the Company's or any Sellers' covenants hereunder;

(xx) enter into any Contract, transaction or arrangement with any Affiliate of the Company;

(xxi) take any action which could have a Material Adverse Effect; or

(xxii) agree, whether in writing or otherwise, to do any of the foregoing.

8.2 <u>Maintenance of Personnel</u>. During the period from the date of this Agreement to the Closing Date, the Company and any Subsidiary agree to cooperate and provide adequate personnel to permit the conduct of the activity contemplated in Section 8.1(a)(i), but only to the extent the Company and its Subsidiary has adequate funding therefor. No Seller shall have any personal liability or obligation to pay for or provide personnel or otherwise fund any activity of the Company, including any activity listed in Section 8.1(a).

8.3 <u>Consent of Governmental Authorities and Others</u>. Each of Parent and Buyer, on the one hand, and the Company or any Subsidiary and Principal Sellers, on the other, agree to file, submit or request (or cause to be filed, submitted or requested) promptly after the Effective Date and to prosecute diligently any and all (a) applications or notices required to be filed or submitted to any governmental or regulatory authorities, and (b) in the case of the Company or any Subsidiary, requests for consents and approvals of Persons, in each case as required to be obtained in connection with the transactions contemplated by this Agreement. Each of Parent and Buyer, on the one hand, and the Company or any Subsidiary and Principal Sellers on the other, shall promptly make available to the other or to a relevant governmental authority, as the case may be, such information as each of them may reasonably request relative to its business, assets and property as may be required by each of them to prepare and file or submit such applications and notices and any additional information requested by any governmental authority, and shall update by amendment or supplement any such information given in writing. Each of Parent and Buyer, on the one hand and the Company or any Subsidiary and Principal Sellers on the other, represent and warrant to the other that such information, as amended or supplemented, shall be true in all material respects and not misleading.

8.4 <u>Due Diligence Review</u>. The Company and any Subsidiary shall (and shall cause its directors, officers, employees, auditors, counsel and agents to) afford Parent and Parent's officers, employees, auditors, counsel and agents reasonable access at all reasonable times to its properties, offices, and other facilities, to its officers and employees, and to all books and records (other than any of the foregoing that constitute information subject to the attorney-client privilege), and shall furnish such persons with all financial, operating and other data and information as may be requested.

8.5 Exclusivity. Except for the transactions contemplated by the Transaction Documents, unless and until this Agreement shall have been terminated, Sellers will not (and Sellers will not cause or permit the Company or any Subsidiary to) (i) solicit, initiate, or encourage the submission of any proposal or offer from any Person relating to the acquisition of any shares or other voting securities, or any substantial portion of the assets, of either Company (including any acquisition structured as a merger, consolidation, or share exchange), or (ii) participate in, or facilitate in any other manner any effort or attempt by any Person to do or seek any of the foregoing or enter into any agreement related to any of the foregoing. Sellers will notify Parent immediately if any Person makes any indication of interest, proposal, offer, inquiry, or contract with respect to any of the foregoing.

8.6 Debt. Parent shall, upon Closing, issue to De Gregorio, 110,000 shares of Parent Common Stock in full satisfaction and discharge of the \$110,000 loan advanced by De Gregorio to the Company (the "Loan Conversion Shares"), and De Gregorio agrees to accept such shares of Parent Common Stock in full satisfaction and discharge of such loan.

8.7 <u>Escrow Agreement</u>. At the Closing, Sellers, Parent, Buyer and the Escrow Agent shall execute and deliver the Escrow Agreement in form and substance reasonably acceptable to the parties.

8.8 Notice of Developments. During the period from Effective Date to the Closing Date, each Party will give prompt written notice

after discovery thereof to the others of any material adverse development that would cause any of such Party's representations or warranties set forth herein to be untrue. No disclosure by any party pursuant to this Section 8.8, however, shall be deemed to amend or supplement the Disclosure Schedules or to prevent or cure any misrepresentation or breach of warranty.

ARTICLE 9

CONDITIONS PRECEDENT; TERMINATION

9.1 <u>Conditions Precedent to the Obligations of Parent and Buyer</u>. The obligations of Parent and Buyer to consummate the transactions contemplated by this Agreement are subject to the satisfaction at or prior to the Closing of the following conditions:

(a) <u>Representations and Warranties True</u>. The representations and warranties of the Sellers and the Company contained in this Agreement, as excepted in the relevant Disclosure Schedules, shall be true and correct in all material respects (except for representations and warranties which are by their terms qualified by materiality, which shall be true and correct to the extent of such materiality) as of the Closing Date with the same force and effect as though made on and as of such date and shall have been true as of the date hereof.

(b) <u>Covenants Performed</u>. The covenants of the Sellers and the Company contained in this Agreement to be performed or complied with on or before the Closing Date shall have been duly performed or complied with.

(c) <u>No Material Adverse Effect</u>. There shall not have occurred any Material Adverse Effect, the impact of which the parties have not been able to resolve to the satisfaction of the parties, acting in good faith and in a commercially reasonable manner.

(d) <u>Company's Certificate</u>. The Company shall have delivered to Buyer a certificate executed by an authorized representative of the Company, on behalf of the Company, dated the Closing Date, certifying in such detail as Buyer may reasonably request, that the conditions specified in this Section have been fulfilled.

(e) <u>No Litigation</u>. No litigation, arbitration or other proceeding shall be pending or threatened against the Company before any court, arbitration panel or governmental authority; no law or regulation shall have been enacted after the date of this Agreement; and no judicial or administrative decision shall have been rendered; in each case, which enjoins, prohibits or materially restricts, or seeks to enjoin, prohibit or materially restrict, the consummation of the transactions contemplated by this Agreement.

(f) <u>Consents</u>. Parent, Sellers and the Company or any Subsidiary shall have obtained all authorizations, waivers, consents and approvals of, and made all filings, applications and notices with, Persons which are necessary or advisable to consummate the transactions contemplated by this Agreement, each of which shall have been obtained without the imposition of any adverse term or condition.

(g) <u>Organizational Documents</u>. The Sellers shall have delivered to Parent certified Organizational Documents and good standing certificates for the Company or any Subsidiary.

(h) Escrow Agreement. The Sellers and the Escrow Agent shall have executed and delivered to Parent the Escrow Agreement.

(i)<u>Delivery of Audited Financial Statements; Closing Trial Balance</u>. The Sellers shall have delivered the Financial Statements and the Closing Trial Balance.

(j) <u>Employment.</u> Giuseppe Santangelo and Buyer shall have agreed upon a written employment agreement to be executed by Skypersonic and Santangelo upon Closing (the "<u>Executive Employment Agreement</u>") providing for the continued employment of Santangelo by Skypersonic in an executive level position, mutually acceptable compensation and benefits and appropriate management incentives, including: (i) incentive stock or option awards for shares of Parent Common Stock; and (ii) annual bonuses upon meeting mutually acceptable benchmarks related to revenue growth and intellectual property goals, payable in cash (or, upon mutual agreement of Santangelo and Buyer, in Parent Common Stock or a combination thereof), protection from disclosure of confidential information, protection of intellectual property and trade secrets, compliance with law, and non-competition substantially on the terms applicable to Sellers herein, for a minimum term of 12 months.

(k) <u>Satisfactory Due Diligence</u>. The Parent shall have completed and be satisfied in all respects with, the results of their ongoing due diligence investigation of the business, assets, operations, properties, financial condition, contingent liabilities, prospects and material agreements of the Company.

(1) <u>Liquidity Event Agreements</u>. The SAFE Holders shall have executed and delivered the Liquidity Event Agreements (other than WSU in the event that WSU has not completed its review and approvals required to execute instruments of conveyance and a Liquidity Event Agreement).

(m) <u>Disclosure Schedules and Exhibits</u>. Within seven (7) days after the Effective Date, Principal Sellers shall have delivered the Disclosure Schedules to Parent, the Parties shall have agreed on the form and substance of all Exhibits to this Agreement, and Parent shall have given to Principal Sellers a written notice of its receipt and approval of the Disclosures Schedules.

9.2 <u>Conditions Precedent to the Obligations of the Sellers</u>. The obligations of the Company and Sellers to consummate the transactions contemplated by this Agreement are subject to the satisfaction at or prior to the Closing of the following conditions:

(a) <u>Representations and Warranties True</u>. The representations and warranties of Parent and Buyer contained in this Agreement or in any certificate or other document delivered pursuant to this Agreement shall be true and correct in all material respects (except for representations and warranties which are by their terms qualified by materiality, which shall be true and correct to the extent of such materiality) as of the

Closing Date with the same force and effect as though made on and as of such date and shall have been true as of the date hereof.

(b) <u>Covenants Performed</u>. The covenants of Parent and Buyer contained in this Agreement to be performed or complied with on or before the Closing Date shall have been duly performed or complied with.

(c) <u>No Material Adverse Effect</u>. There shall not have occurred any Material Adverse Effect, the impact of which the parties have not been able to resolve to the satisfaction of the parties, acting in good faith and in a commercially reasonable manner.

(d) <u>No Litigation</u>. No litigation, arbitration or other proceeding shall be pending or threatened against Parent or Buyer before any court, arbitration panel or governmental authority; no law or regulation shall have been enacted after the date of this Agreement; and no judicial or administrative decision shall have been rendered; in each case, which enjoins, prohibits or materially restricts, or seeks to enjoin, prohibit or materially restrict, the consummation of the transactions contemplated by this Agreement.

(e) <u>Parent's Certificate</u>. Parent shall have delivered to the Sellers a certificate executed by an authorized officer of Parent dated the Closing Date certifying in such detail as Sellers may reasonably request that the conditions specified in this Section have been fulfilled.

(f) Escrow Agreement. Parent, Buyer and Escrow Agent shall have executed and delivered to the Sellers the Escrow Agreement.

(g) <u>Executive Employment Agreement</u>. The execution and deliver by Skypersonic to Santangelo of the Executive Employment Agreement at Closing.

(h) Satisfactory Due Diligence. Sellers shall have completed and be satisfied with their due diligence review of Parent and Buyer.

(i)<u>Consents</u>. Sellers and the Company shall have obtained all authorizations, waivers, consents and approvals of, and made all filings, applications and notices with, Persons which are necessary or advisable to consummate the transactions contemplated by this Agreement, including, without limitation, approval of this Agreement and the transactions hereunder by the Board of Directors of Skypersonic and either (i) the election by WSU to waive the cash option under its SAFE Agreement and agree to accept its proportionate share of the Stock Consideration in full satisfaction of its rights under its SAFE Agreement or (ii) the execution of a written agreement between WSU and Parent pursuant to which Parent or Buyer agrees to pay an amount of cash to WSU to fully pay the amount of the cash option under the SAFE Agreement and WSU's elects to accept payment of such cash in satisfaction of its rights under the SAFE Agreement.

(j)Loan Conversion Shares. Parent shall have issued the Loan Conversion Shares at Closing.

(k) Listing. Parent's Common Stock shall either remain traded on the OTCQB or the OTCQX or is approved for listing on the NASDAQ Capital Market.

(1)<u>Trading Restrictions</u>. None of the Stock Consideration shall be subject to any lock-up agreement or other contractual restriction on disposition (other than the Escrow Agreement).

(m)<u>Closing Price</u>. On the trading day immediately prior to the Closing Date, the closing price per share of the Parent Common Stock shall be not less than Two and 50/100 Dollars (\$2.50), unless Parent and Buyer agree to increase the Purchase Price by issuing additional shares to Sellers.

(n) <u>34 Act Reporting</u>. Parent shall be subject to the reporting requirements under Section 13 or 15(d) of the Securities Exchange Act of 1934, as amended, and Parent shall have timely filed all reports required thereunder during the period following the Effective Date.

(o) Liquidity Event Agreements. The SAFE Holders shall have executed and delivered the Liquidity Event Agreements.

(p) <u>Disclosure Schedules and Exhibits</u>. Within seven (7) days after the Effective Date, Principal Sellers shall have delivered the Disclosure Schedules to Parent, the Parties shall have agreed on the form and substance of all Exhibits to this Agreement, and Parent shall have given to Principal Sellers a written notice of its receipt and approval of the Disclosures Schedules.

9.3 Termination.

(a) Anything herein or elsewhere to the contrary notwithstanding, this Agreement may be terminated at any time before the Closing Date only as follows:

(i) by mutual consent of the Sellers and Parent or if Closing is prohibited by change in Law;

(ii) by Parent, if there has been a material breach by any Sellers at any time before the Closing of any representation, warranty, covenant or agreement contained in this Agreement that is not curable or, if curable, is not cured within thirty days or such other time as may be agreed by the parties in writing after written notice of such breach is given by Parent to the party committing such breach;

(iii) by any Sellers if there has been a material breach by Parent or Buyer at any time before the Closing of any representation, warranty, covenant or agreement contained in this Agreement that is not curable or, if curable, is not cured within thirty days after written notice of such breach is given by such Sellers to the party committing such breach;

(iv) by Parent, on the one hand, or any Sellers, on the other hand, upon written notice to all Parties if the conditions specified in Section 9.1(m) or Section 9.2(p) have not been satisfied within the time period specified therein;

(v) by Parent, on the one hand, or any Sellers, on the other hand, upon notice given to all parties if the Closing shall not have taken place within ninety (90) days of the Effective Date; or

(vi) by Parent, on the one hand, or any Sellers, upon notice given to all parties if any court (i) shall have issued an order, decree or ruling or taken any other action permanently restraining, enjoining or otherwise prohibiting the transactions contemplated by this Agreement, and such order, decree, ruling or other action shall have become final and nonappealable or (ii) shall have failed to issue an order, decree or ruling or to take any other action, as applicable, and such denial of a request to issue such order, decree, ruling or take such other action shall have become final and nonappealable, in the case of each of (i) and (ii) which is necessary to fulfill the conditions set forth in Article 9; provided, however, that the right to terminate this Agreement under this Section 9.3 shall not be available to any party whose failure to comply with Section 8.3 has been the cause of such action or inaction.

(b) In the event of the termination of this Agreement as provided in Section 9.3(a), this Agreement shall forthwith become wholly void and of no further force and effect (except as set forth in this Section 9.3, Sections 7.4, 7.5, 10.7, 10.11, 10.12, 10.13, 10.14 and 10.15).

9.4 <u>No Waiver</u>. Notwithstanding anything in this Agreement to the contrary, in the event that Parent consummates the transactions contemplated hereunder and effectuates the Closing, such actions shall in no way and at no time be considered a waiver or release of any breach of any representation, warranty or covenant of any Sellers or any Company, including, without limitation, the right to indemnification pursuant to Article 7.

ARTICLE 10

MISCELLANEOUS

10.1 Notices. Any notice or other communication under this Agreement shall be in writing and shall be delivered personally or sent by certified mail, return receipt requested, postage prepaid, or sent by email or facsimile or prepaid overnight courier to the parties at the addresses set forth below their names on the signature pages of this Agreement (or at such other addresses as shall be specified by the parties by like notice). Such notices and other communications shall be deemed given when actually received or (a) in the case of delivery by overnight service with guaranteed next day delivery, the next business day after deposited for next business day delivery or the day designated for delivery, (b) in the case of email, the date of transmission if the recipient acknowledges receipt thereof in writing (including by reply email), (c) in the case of facsimile, the date upon which the transmitting party received confirmation of receipt by facsimile, telephone or otherwise. A copy of any notices delivered to Parent shall also be sent to: Red Cat Skypersonic, Inc. Attn: Jeffrey Thompson, CEO, Jeff@Redcat.red. A copy of any notices delivered to the Sellers, or to the Company prior to the Closing, shall also be sent to John A. Sellers, Esq., Sellers, P.C., 32400 Telegraph Road, Suite 104, Bingham Farms, Michigan 48025, Fax (248) 723-1632, Email: jsellers@sellerspc.com.

10.2 <u>Entire Agreement</u>. This Agreement, together with the Escrow Agreement, its schedules and exhibits contain every obligation and understanding between the parties relating to the subject matter hereof, and merges all prior discussions, negotiations and agreements, if any, between them as to the same subject matter, and none of the parties shall be bound by any representations, warranties, covenants, or other understandings, other than as expressly provided or referred to herein or therein.

10.3 <u>Assignment</u>. This Agreement may not be assigned by any party without the written consent of the other parties. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors, heirs, personal representatives, legal representatives, and permitted assigns.

10.4 <u>No Third Party Beneficiary</u>. Nothing expressed or implied in this Agreement is intended, or shall be construed, to confer upon or give any Person other than the parties hereto and their respective heirs, personal representatives, legal representatives, successors and permitted assigns, any rights or remedies under or by reason of this Agreement.

10.5 Waiver and Amendment. Any representation, warranty, covenant, term or condition of this Agreement which may legally be waived, may be waived at any time by the party entitled to the benefit thereof, and any term, condition or covenant hereof may be amended by the parties hereto at any time by written agreement. Any such waiver or amendment shall be evidenced by an instrument in writing executed on behalf of the appropriate party by a person who, to the extent applicable, has been authorized by its board of directors to execute waivers, extensions or amendments on its behalf. No waiver by any party hereto, whether express or implied, of its rights under any provision of this Agreement shall constitute a waiver of such party's rights under such provisions at any other time or a waiver of such party's rights under any other provision of this Agreement. No failure by any party hereto to take any action against any breach of this Agreement or default by another party shall constitute a waiver of the former party's right to enforce any provision of this Agreement or to take action against such breach or default or any subsequent breach or default by such other party.

10.6 <u>Severability</u>. In the event that any one or more of the provisions contained in this Agreement shall be declared invalid, void or unenforceable, the remainder of the provisions of this Agreement shall remain in full force and effect, and such invalid, void or unenforceable provision shall be interpreted as closely as possible to the manner in which it was written.

10.7 Expenses. Each party agrees to pay, without right of reimbursement from the other party, the costs (hereafter referred to as "<u>Costs</u>") incurred by it incident to the performance of its obligations under this Agreement and the consummation of the transactions contemplated hereby, including, without limitation, costs incident to the preparation of this Agreement, and the fees and disbursements of counsel, accountants and consultants employed by such party in connection herewith. The Costs of counsel and accountants for Skypersonic shall remain obligations of, and shall be paid by, Skypersonic. The Sellers covenant to Parent that in no event shall any Costs of Sellers be paid by Skypersonic.

10.8 Headings and References. The section and other headings contained in this Agreement are for reference purposes only and shall

not affect the meaning or interpretation of any provisions of this Agreement. References in this Agreement to clauses, subclauses, sections, articles or schedules are references to clauses, subclauses, sections, articles or schedules of this Agreement so numbered.

10.9 <u>Time of Essence</u>. With regard to all dates and time periods set forth or referred to in this Agreement, time is of the essence.

10.10 <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (including facsimile or PDF), each of which shall be deemed an original but all of which together shall constitute one and the same instrument.

10.11 Litigation; Prevailing Party. In the event of any litigation with regard to this Agreement, the prevailing party shall be entitled to receive from the non-prevailing party and the non-prevailing party shall pay upon demand all reasonable fees and expenses of counsel for the prevailing party; provided, however, to the extent that both parties prevail on claims filed against the other, in a consolidated action, then the aggregate amount of fees and expenses of counsel to both parties shall be allocated between Parent and the Sellers so that each party's aggregate share of such fees and expenses bears the same proportion to the total amount of such fees and expenses as the amount awarded each party on their prevailing claim bears to the total amount awarded in the litigation on all prevailing claims and any party that has paid less than its allocated expenses shall immediately remit to the other party the difference between the amount allocated and the amount previously paid.

10.12 <u>Governing Law</u>. This Agreement has been entered into and shall be construed and enforced in accordance with the laws of the State of Nevada, without reference to the choice of law principles thereof.

10.13 Jurisdiction and Venue. This Agreement shall be subject to the exclusive jurisdiction of the U.S. District Court for the District of Nevada. The parties to this Agreement agree that any breach of any term or condition of this Agreement shall be deemed to be a breach occurring in the State of Nevada by virtue of a failure to perform an act required to be performed in the State of Nevada and irrevocably, unconditionally and expressly agree to submit to the jurisdiction of the U.S. District Court for the District of Nevada sitting in Clark County, Nevada for the purpose of resolving any disputes among the parties relating to this Agreement or the transactions contemplated hereby. The parties irrevocably waive, to the fullest extent permitted by law, any objection or immunities to jurisdiction which they may now or hereafter have (including sovereign immunity, immunity to pre-judgment attachment, post-judgment attachment and execution) to the laying of venue of any suit, action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, or any judgment entered by any court in respect hereof brought in the State of Nevada, and further irrevocably waive any claim that any suit, action or proceeding brought in Nevada has been brought in an inconvenient forum. To the extent that any party hereto has or hereafter may acquire any immunity from jurisdiction or otherwise) with respect to itself or its property, it hereby irrevocably waives such immunity in respect of its obligations under this Agreement and the transactions contemplated hereby to the extent permitted by law.

10.14 <u>Publicity</u>. The parties shall agree to the content of any press release or other public announcement concerning this Agreement or the transactions contemplated hereby before issuing the same. Nothing contained herein shall prevent any party from at any time furnishing any information to any governmental authority which it is by law or otherwise so obligated to disclose or from making any disclosure which its counsel deems necessary or advisable in order to fulfill such party's disclosure obligations under applicable U.S. law or the rules of the any stock exchange to which the party is subject.

10.15 <u>Construction</u>. The language in all parts of this Agreement shall in all cases be construed simply, according to its fair meaning, and not strictly for or against any of the parties. Without limitation, there shall be no presumption against any party on the ground that such party was responsible for drafting this Agreement or any part thereof, and any rule of law, or any legal decision that would require interpretation of any claimed ambiguities in this Agreement against the party that drafted it has no application and is expressly waived.

(Signature Page Follows)

SIGNATURE PAGE TO SHARE PURCHASE AGREEMENT

NOW, THEREFORE, the parties hereto have each executed and delivered this Agreement on the date specified next to their respective signatures below, with the intention that that Agreement shall be effective on and as of the Effective Date.

PARENT:

RED CAT HOLDINGS, INC., a Nevada corporation

Dated: February ____, 2021

By: ______ Jeffrey Thompson Its: Chief Executive Officer

Address:

BUYER:

	RED CAT SKYPERSONIC, INC., a Nevada corporation
Dated: February, 2021	By: Jeffrey Thompson Its: Chief Executive Officer Address: [Signatures continue on the next page]
	CONTINUATION OF SIGNATURE PAGE TO SHARE PURCHASE AGREEMENT COMPANY :
	SKYPERSONIC, INC., a Michigan corporation
Dated: February, 2021	By: Giuseppe Santangelo Its: Chief Executive Officer
	<u>SELLERS</u> :
Dated: February, 2021	Giuseppe Santangelo, individually
Dated: February, 2021	Anna Piccione, individually
Dated: February, 2021	Michele De Gregorio, individually
Dated. redruary, 2021	Michele De Gregorio, individually

DISCLOSURE SCHEDULES

See attached.

EXHIBIT A

(Form of Escrow Agreement)

See attached.

LIQUIDITY EVENT AGREEMENT

The undersigned ("*Shareholder*") is a holder of shares of common stock (the "*Shares*") of Skypersonic, Inc., a Michigan corporation (the "*Company*"). In connection with a proposed purchase by Red Cat Skypersonic, Inc, a Nevada corporation ("*Purchaser*"), an affiliate of Red Cat Holdings, Inc., a Nevada corporation ("*Red Cat*"), of 100% of the issued and outstanding capital stock of the Company in exchange for newly issued shares of Red Cat (the "*Transaction*"), Shareholder and the Company hereby agree as follows:

1. The closing of the contemplated Transaction will constitute a Liquidity Event, as such term in defined in certain SAFE Agreements (the "SAFE Agreements") between the Company and various investors identified on Exhibit A attached hereto (the "SAFE Investors").

2. Notwithstanding any provision of the SAFE Agreements to the contrary concerning the amount or the shares of common stock of the Company to which the SAFE Investors would be entitled to receive upon the occurrence of a Liquidity Event, for Shareholder's Shares upon the closing of the Transaction, Shareholder agrees to accept in the Transaction Shareholder's proportionate share of the total Transaction consideration to be paid by Red Cat at or in connection with the Closing, as allocated to Shareholder on Exhibit A attached hereto. In the event the purchase price to be paid by the Purchaser in the Transaction is reduced to an amount that is less than 3,000,000, the amount of such reduction shall be allocated among those who are currently holders of the Company's common stock and those who are currently holders of all SAFE Agreements in accordance with the Ultimate Sharing Ratios specified in Exhibit A

3. Shareholder acknowledges and agrees that 50% of the Transaction consideration will be deposited into escrow for a period of one (1) year following the closing as security for any indemnification claims which may be asserted by Red Cat or Purchaser based on any breach or violation of any of the representations and warranties contained in the Share Purchase Agreement. Shareholder further agrees that the number of any escrowed shares which are forfeited as a result of a successful indemnification claim by Red Cat or Purchaser based on a breach or violation of any representation or warranty of or concerning the Company contained in the Share Purchase Agreement shall be allocated among those who are currently holders of the Company's common stock and those who are currently holders of all SAFE Agreements in accordance with the Ultimate Sharing Ratios specified in Exhibit A.

4. If the Transaction does not close, this Liquidity Event Agreement shall terminate and Shareholder's rights shall continue unaffected by any provision hereof.

SHAREHOLDER:

COMPANY:

Skypersonic, Inc., a Michigan corporation

Giuseppe Santangelo

Dated: February ____, 2021

By: _____ Giuseppe Santangelo

Its: Chief Executive Officer

Dated: February ____, 2021

SENIOR SECURED PROMISSORY NOTE

\$75,000.00

FOR VALUE RECEIVED, **SKYPERSONIC**, **INC.**, a Michigan corporation ("<u>Maker</u>") promises to pay **Fat Shark Holdings**, **Ltd.**, a Cayman Islands Exempted Company, or assigns ("<u>Holder</u>") in lawful money of the United States of America, the aggregate principal sum of Seventy-Five Thousand Dollars and 00/100 (\$75,000.00), together with interest on the unpaid principal balance from the date hereof until the Payment Date (as defined below) at an annual rate equal to six (6%) percent per annum, in the manner provided below. Interest shall be calculated on the basis of a year of 365 or 366 days, as applicable, and charged for the actual number of days elapsed until payment.

1. <u>Principal and Interest</u>. The principal amount of this Note and the unpaid interest shall be due and payable on May 14, 2021 (the "<u>Payment Date</u>"). Notwithstanding anything herein to the contrary, the Note may be prepaid by Maker without penalty, in whole or in part. In the event of the sale of any of the Collateral (as defined below) by Maker, the Payment Date with respect to the net proceeds of such sale shall be accelerated to the date of receipt by Maker and shall be immediately paid to Holder as a reduction in the Note obligations.

2. <u>Manner of Payment</u>. The payment of principal and interest on this Note shall be paid by Maker to Holder by wire transfer of immediately available funds to an account or accounts designated by Holder in writing. If any payment of principal or interest on this Note is due on a day which is not a Business Day, such payment shall be due on the next succeeding Business Day, and such extension of time shall be taken into account in calculating the amount of interest payable under this Note. "<u>Business Day</u>" means any day other than a Saturday, Sunday or legal holiday in the State of New York.

3. <u>Security</u>. Maker's performance under this Note is secured by 54,250 shares of common stock of Maker owned of record by Giuseppe Santangelo ("Santangelo"). an affiliate of Maker (the "<u>Collateral</u>") which are hereby pledged as collateral to Holder for the obligations herein in the event of a default in payment or performance of any of Maker's obligations under this note. Upon the closing of a sale of the outstanding capital stock of Maker under that certain Securities Purchase Agreement dated as of February 11, 2022 (the "SPA"), this Section 3 shall automatically terminate, Santangelo shall have no obligation under this Note, the security interest granted hereunder shall terminate, and Santangelo shall transfer, sell and convey the Collateral to Buyer in consideration for the Purchase Price as set forth therein, and neither Holder nor any assignee of Holder shall have any rights or interests of any nature in the Collateral or any proceeds from the sale of the Collateral by Santangelo to Buyer for the pro-rata portion of the Purchase Price, as set forth therein.

4. <u>SPA Provisions</u>. Terms not otherwise defined herein shall have the meanings ascribed to such terms in the SPA. All obligations under the Note, for the absence of doubt, including principal, interest, penalties and costs, shall be deemed "current liabilities" in the calculation of Closing Date Working Capital.

5. <u>Prepayment</u>. Maker may, without premium or penalty, at any time and from time to time, prepay all or any portion of the outstanding principal balance due under this Note, provided that each such prepayment is accompanied by accrued interest on the amount of principal prepaid calculated to the date of such prepayment.

6. <u>Severability</u>. If any provision in this Note is held invalid or unenforceable by any court of competent jurisdiction, the other provisions of this Note will remain in full force and effect. Any provision of this Note held invalid or unenforceable only in part or degree will remain in full force and effect to the extent not held invalid or unenforceable.

7. Governing Law. This Note will be governed by the laws of the State of New York without regard to principles of conflicts of laws.

8. <u>Parties in Interest</u>. This Note may be assigned or transferred by Holder. Subject to the preceding sentence, the rights and obligations of Maker and Holder shall be binding upon and benefit their successors, assigns, heirs, administrators and transferees.

9. Section Headings; Construction. The headings of Sections in this Note are provided for convenience only and will not affect its construction or interpretation. All references to "Section" or "Sections" refer to the corresponding Section or Sections of this Note unless otherwise specified. All words used in this Note will be construed to be of such gender or number as the circumstances require. Unless otherwise expressly provided, the words "hereof" and "hereunder" and similar references refer to this Note in its entirety and not to any specific section or subsection hereof.

10. <u>Representation</u>. Each party hereto acknowledges that it has been represented by independent legal counsel in the preparation of the Agreement. Each party recognizes and acknowledges that the undersigned has served as counsel to the Maker and may, in the future, represent the Maker in connection with various legal matters and each party waives any conflicts of interest including claims related to the adequacy of consent and of legal services and other allegations that it has not been represented by its own counsel.

11. Exclusivity. Except for the transactions contemplated by the Transaction Documents and hereunder, Santangelo will not (i) solicit, initiate, or encourage the submission of any proposal or offer from any Person relating to the acquisition of any shares or other voting securities, or any substantial portion of the assets, of either Company (including any acquisition structured as a merger, consolidation, or share exchange), or (ii) participate in, or facilitate in any other manner any effort or attempt by any Person to do or seek any of the foregoing or enter into any agreement related to any of the foregoing. Sellers will notify Parent immediately if any Person makes any indication of interest, proposal, offer, inquiry, or contract with respect to any of the foregoing. Santangelo agrees, on his own behalf and on behalf of Maker, that neither Santangelo nor Skypersonic shall have the right to terminate the SPA prior to the Payment Date.

IN WITNESS WHEREOF, Maker has executed and delivered this Note as of the date first written above.

SKYPERSONIC, INC.

By:______ Name: GIUSEPPE SANTANGELO Title: Chief Executive Officer

GIUSEPPE SANTANGELO

Individually with respect to paragraph 3 hereof

ACKOWLEDGMENT AND CONSENT:

Red Cat Holdings, Inc., a Nevada corporation ("<u>Red Cat</u>"), and Red Cat Skypersonic, Inc., a Nevada corporation ("<u>Stock Purchaser</u>"), are parties to a certain Share Purchase Agreement (the "<u>Purchase Agreement</u>") pursuant to which Stock Purchaser agreed to purchase all of the issued and outstanding capital stock of Maker, including all shares of common stock of Maker held by Giuseppe Santangelo, in exchange for newly issued shares of the common stock of Red Cat. Red Cat and Stock Purchaser hereby: (a) consent to the foregoing Note, the indebtedness evidenced thereby, and to the pledge by Giuseppe Santangelo of the Collateral to secure such indebtedness; and (b) agree that all applicable representations and warranties by or on behalf of Maker and Giuseppe Santangelo in the Purchase Agreement and the Disclosure Schedules thereto are amended and supplemented to include this Note, the indebtedness hereunder and the pledge of the Collateral.

RED CAT HOLDINGS, INC., a Nevada corporation

Dated: February ____, 2021

By: ______ Jeffrey Thompson Its: Chief Executive Officer

RED CAT SKYPERSONIC, INC., a Nevada corporation

Dated: February ____, 2021

By: _____ Jeffrey Thompson Its: Chief Executive Officer Red Cat Signs Definitive Agreement to Acquire Skypersonic and its "Fly Anywhere" Drone Technologies



ORLANDO, FL – February 16, 2021 (PRNewswire) – Red Cat Holdings, Inc. (OTC: RCAT) ("Red Cat" or the "Company"), a leading brand in the drone industry, announced the signing of a definitive agreement to acquire Skypersonic, Inc., a provider of drone products and software solutions that enable drone inspection flights that can be executed by pilots anywhere in the world.

Skypersonic powers drones to "Fly Anywhere" and "Inspect the Impossible". Its patented software and hardware solutions allow for inspection services in restricted spaces where GPS is not allowed or available. Skycopter is a miniature drone fitted into a cage to avoid damage to inspected areas and the drone. Skyloc is a stand-alone, real time, software system which enables the drone to record and transmit inspection data while being operated from thousands of miles away. Skypersonic's intellectual property portfolio includes eight US and European patents.

"The addition of Skypersonic will further strengthen our growing suite of highly specialized and unique drone services," stated Jeff Thompson, CEO of Red Cat. "We continue to build a strong platform of products and services in the drone industry which is forecast to reach \$40 billion by 2025." (Globe Newswire)

"Skypersonic currently services global customers in the automotive and energy industries who rely on our drone platform to perform real-time inspections in remote locations using pilots and inspectors located all over the world." stated Giuseppe Santangelo, CEO of Skypersonic. "Combining our talent and resources with Red Cat will enable us to accelerate growth in established markets while expanding our reach into new territories."

About Red Cat

Red Cat provides products, services and solutions to the drone industry through its three wholly owned subsidiaries. Fat Shark Holdings is the leading provider of First Person View (FPV) video goggles to the drone industry. Rotor Riot, LLC is a leader in the sale of FPV drones and equipment, primarily to the consumer marketplace through its digital storefront located at <u>www.rotorriot.com</u>. Rotor Riot enjoys high visibility in social media through its Facebook page and its sponsorship of a professional drone racing team which has won numerous championships. Red Cat Propware is developing a Software-as-a-Solution ("SaaS") platform to provide drone flight data analytics and storage, as well as diagnostic products and services.

About Skypersonic Inc.

Skypersonic Inc., is a provider of drone products and software solutions designed to "Fly Anywhere" and "Inspect the Impossible". Its patented software and hardware solutions allow for inspection services in restricted spaces where GPS is not allowed or available. Skycopter is a miniature drone fitted into a cage to avoid damage to inspected areas and the drone. Skyloc is a stand-alone, real time, software system which enables the drone to record and transmit inspection data while being operated from thousands of miles away. Skypersonic's intellectual property portfolio includes eight US and European patents.

Safe Harbor

The information provided in this press release may include forward-looking statements relating to future events or the future financial performance of the Company. Because such statements are subject to risks and uncertainties, actual results may differ materially from those expressed or implied by such forward-looking statements. Words such as "anticipates," "plans," "expects," "intends," "will," "potential," "hope" and similar expressions are intended to identify forward-looking statements. These forward-looking statements are based upon current expectations of the Company and involve assumptions that may never materialize or may prove to be incorrect. Actual results and the timing of events could differ materially from those anticipated in such forward-looking statements as a result of various risks and uncertainties. Detailed information regarding factors that may cause actual results to differ materially from the results expressed or implied by statements in this press release relating to the Company may be found in

the Company's periodic filings with the Securities and Exchange Commission, including the factors described in the sections entitled "Risk Factors", copies of which may be obtained from the SEC's website at <u>www.sec.gov</u>. The parties do not undertake any obligation to update forward-looking statements contained in this press release.

Contact

Chad Kapper Phone: (818) 906-4701 E-mail: <u>support@rotorriot.com</u> Website: <u>https://rotorriot.com</u>

Investor Relations Contact Bruce Haase RedChip Companies (407) 712-8965 bruce@redchip.com



CONFIDENTIAL NOT FOR DISTRIBUTION © 2021 RED CAT HOLDINGS, All Rights Reserved February 2021 : RCAT INVESTOR PRESENTATION

2

DISCLOSURES

During the course of this presentation, we may make statements which are considered "forward-looking statements" within the meaning of the Private Securities Litigation. Reform Act of 1995, including statements regarding our future performance, goals and growth. Forward-looking statements may be identified by such words as "believe," "estimate," "anticipate," "expect," "plan," "intend," "may," "could," "might," will," "should, "approximately," "potential" or, in each case, their negative or other variations thereon or comparable terminology as it relates to us, although not all forward-looking statements contain these words.

We have based the forward-looking statements largely on our current expectations and projections about the future events and financial trends that we believe may affect our financial condition, results of operations, business strategy and financial needs. Important factors that could cause actual results to differ from those in the forward-looking statements include inability to successfully integrate the assets acquired in the recently closed merger and asset purchase and the Risk Factors contained within our filings with the Securities and Exchange Commission including our annual report on Form 10–Q for the period ended March 31, 2019. Any forward-looking statement made by us herein speaks only as of the date on which it is made. We undertake no obligation to publicly update any forward-looking statement, whether as a result of new information, future developments or otherwise, except as may be required by Iaw.

RED CAT HOLDINGS : RCAT

INVESTMENT HIGHLIGHTS

THROUGH OUR FOUR OPERATING SUBSIDIARIES, WE PROVIDE drone products, services and technologies to the fastest growing market segments in our industry In 2020, we created RED CAT Drone Services, a dedicated drone services company that combines the capabilities and technologies in our portfolio holdings in one comprehensive solution

- Uniquely positioned to capitalize on the maturing drone market with high performance products and business services for a variety of industries.
- Disciplined acquisition strategy targets companies with advanced product offering and specialized drone operational platforms and IP
- Experienced management team with successful financial and technology background



RCAT EXECUTIVE TEAM

JEFFREY THOMPSON | Chief Executive Officer and Chairman of the Board is a serial tech entrepreneur with a long track record of founding and leading companies from the startup phase to successful exits and IPOs. His background is in the telecom space, where he founded EdgeNet, a private Intermet Service provider (ISP), which was sold to Citadel Broadcasting in 1997. Thompson then co-founded Towestream, a wireless service provider (public 2007 NASDAQ). With an ongoing passion for innovative technology, he most recently founded. Red Cat, the company at the forefront of providing solutions for the rapidly growing commercial and recreational drone industries. industries.

JOSEPH HERNON | Chief Financial Officer is a Certified Public Accountant with a Master's in Accountancy, from Bentley University. He also has experience being a member of management teams that have raised more than \$300 million through debt and equity financings. As a Chief Financial Officer of 4 publicly traded companies in the high tech and biotech industries, Joseph led S acquisitions of technology companies which were immediately accretive.

MANAGEMENT TEAM

ALLAN EVANS | FAT SHARK President is a serial entrepreneur with a history of founding and leading technology innovation. He received his PhD from the University of Michigan in Electrical Engineering. From there, PhD throm the University of Michigan in Electrical Engineering. From there, he went to Pacific Northwest National Labs and eventually spun out one of the technologies he developed into Avegant, a VR/AR company. After Avegant, Allan became the CEO of Fat Shark and profitably managed the company which was recently combined with Red Cat. Allan has a passion for consumer technologies and leading rapidly growing technology team in emerging markets.

CHAD KAPPER | ROTOR RIOT President has more than 25 years of media and marketing experience, Kapper created two of the most successful brands in the drone industry. Flite Test and Rotor Riot have almost 2 million subscribers and the content has been viewed more than 300,000,000 times worldwide

4

RED CAT HOLDINGS : RCAT





COMMERCIAL AND CIVIL APPLICATIONS

The use cases for drones in commercial and civil services are focused on the need to INSPECT and/or SURVEY equipment, structures, insurance claims, land areas, jobsites and emergencies

Addressable Markets for Drone Inspection and Survey Services 5E 0 D



for FIRST RESPONDERS...

111.2 B

The Ever-Expanding List of Drone Uses and Applications

for OIL & GAS...



INDUSTRY CHALLENGES CREATED OPPORTUNITIES

CHALLENGES.

- 1. Operators' flight data collection systems vary greatly, and some do not meet regulatory compliance standards and reporting 2. Experienced pilots are in high demand but short supply
- Specific response are in high demand but short supply
 Downlink technologies' high latency limits accuracy and speed to complete operations Remote operation is hampered by technology performance and platform standardization.

OPPORTUNITIES

- CREATE A DRONE SERVICES PLATFORM for operators and with the highest standards in flight data collection, advanced technologies and meets regulatory compliance.
- V DEVELOP A NETWORK OF CERTIFIED PILOTS and learning programs to train and certify pilots.
- ADVANCE DRONE TECHNOLOGIES to enable operations by experienced pilots and professionals from any location with low latency and 5G data transmission.

RED CAT HOLDINGS : RCAT





Click Below to View Our Video



INVESTING IN THE FUTURE OF DRONES

through Products, Technologies and Services Focused on the Market's Fastest Growing Segments and Use Cases

8

SUBSIDIARY BUSINESS LINES AND VALUE

Strategically-Targeted Acquisitions Provided Us Four (4) Business Lines with Dedicated Revenue Streams Plus the Software, Technologies, Services and Exposure Needed to Build a Comprehensive Drone Services Company



FAT SHARK FIRST PERSON VIEW **DRONE TECHNOLOGY**

- Fat Shark is the premium market share leader of first-person view (FPV) headsets for drones
- Highly-advanced download technologies and firmware . produce the lowest latency in the industry : low latency is essential for drone operations
- FatShark headsets are routinely awarded with the ۲ highest performance grades and operator preference in the market
- 🗸 Fat Shark currently sells 100% through distribution at ~26% margins

FAT SHARK® Headset Revenues



RED CAT HOLDINGS : RCAT



SKYPERSONIC INSPECT & SURVEY IN GPS-DENIED LOCATIONS

- Vatented software and hardware allow for flight services in restricted spaces where GPS is not allowed or available
- GPS-denied, indoor/tight space telemetry capabilities were transitioned from military and satellite technology for commercial use - 8 US and European Patents
- 😻 Environmental testing technologies samples air quality
- Provides operators unique access to locations :where previous human operators' high risk = high costs
- ♥ Completely remote drone flights can be executed from operators anywhere in the world
- Drone services, technologies and certification programs offered

Sample Customers Who Contracted Skypersonic Services or Purchased Skypersonic Products



RED CAT HOLDINGS : RCAT

SKYCOPTER INSPECT THE IMPOSSIBLE



EROCK

'FLY ANYWHERE' TECHNOLOGIES

ele drones to inspect and y in high-risk, confined or denied locations by 100%

DRONEBOX FLIGHT LOGGING **& REGULATORY COMPLIANCE**

Red Cat's DRONE BOX SasS software and platform enable an easy-to-use flight log system keeps clients compliant with regulators and helps track critical, drone data collection and feeds to various applications.

- Patent-pending blockchain-based cloud architecture is Drone-agnostic .
- ONE easy to use system for analytics and services
- V Partnerships with largest drone equipment suppliers
- V Intellectual property agreement with UPR Mayagüez
- ♥ Launching AI capabilities
- Verto Rico based operations Cost of operations 30% less than mainland US
- Puerto Rico Tax incentives, ACT 20, 22 and 73
- Total income tax rate of 4%



ROTOR RIGT EXPOSURE AND ACCESS TO TOP PILOTS

- ♥ At the center of the high-performance drone market and has the largest following of professional and 107-Certified drone pilots
- ♥ Enthusiasts shop online store for the ready-to-fly drones and drone parts featured in Rotor Riot's media
- Drone Racing League events and media

WE AVAIL CERTIFIED PILOTS ON A REMOTE-OPERATION PLATFORM

to provide widespread drone services to for commercial and civil service use

RED CAT HOLDINGS : RCAT

♥ A unique mobile game and drone simulation controller is sold on the shelves of Best Buy, Apple Store, Target and Walmart



Strong Social Media I Provide Worl	Following Including dwide Exposure – 20	
• v • i	-	

🔼 YouTube	0	· · ·
215,000	25,000,000	150,000
Subscribers	Views	Social Media Followers

RED CAT HOLDINGS : RCAT



REDCAT DRONE OPERATIONS AS A BUSINESS SERVICE

WHAT WE OFFER

- VHAI WE OFFER
 1. Pilots who are remote, quickly available, experienced and certified to fly drone operations
 2. A platform to access and process drone data collection and maintain compliance
 3. Data for professionals who can recognize and interpret data provided during and after drone inspection and survey services

HOW OUR REMOTE PILOTING AND OPERATIONS BEST SERVICE CLIENTS

- Customers get a cost-effective and safer solution no need to be in-field
- no need to be in-field Access to industry-experienced, certified pilots worldwide Delivery of drone data for specialists and professionals to linterpret data in real-time, or post-operations to help clients make the best decisions. 2. 3.



14

SUBSIDIARIES' CAPABILITIES SPECIALIZE **REDCAT'S DRONE SERVICES OFFERINGS**

Specific capabilities and technologies in each of our subsidiaries contributed to a suite of highly-specialized and unique services in our RED CAT Drone Services division

RCAT Subsidiary	DRONEBOX Ecorpred Decage and Analyzation	ROTOR RIOT	FAT SHARK	SKYPERSONIC
THE CAPABILITY Our Subsidiary Contributed to Drone Services	A SOFTWARE PLATFORM that enables immediate compliance and data feeds for all drone operations	A NETWORK OF PILOTS for us to quickly connect and schedule a professional and certified pilot for our operations	DOWNLINK TECHNOLOGY which enables low-latency, FPV piloted flights over the internet	CPS-DENIED FLIGHT OPERATIONS using digital telemetry for high-risk drone flights by 100% remotely piloted operators
REDCAT				
	to	apabilities we leveraged fi assemble a comprehensi pection and survey service:	ve offering of remote droi	ne



USE CASES FOR OUR DRONE SERVICES DIVISION

CONSTRUCTION on and Survey 0

operations, mopeetion		
	Site Planning	
8	Equipment Operation	

```
    Site Planning
    Equipment Operation and
Safety Inspection
    Structural Integrity
```

INDUSTRIAL

Operations, Inspection and Survey

- Equipment maintenance Pipelines and rigs Cell towers Power networks Power production
- Agricultural Survey
 Confined space
 Unsafe environments

Grading Survey Zoning boundaries Immediate Response

CIVIL SERVICES Operations, Inspection, Survey, First Response

Building Codes
 Infrastructure maintenance
 Road / Bridges / Tunnen
 Zoning / Property Lines
 June data
 Survey

16

RCAT INVESTMENT SUMMARY

- 1. Uniquely positioned to capitalize on the maturing drone market with high performance products, technologies and business services for a variety of industries
- 2. Acquired businesses complement and support each other and add value to REDCAT Drone Services offering
- 3. Newly acquired business and technologies provide unique, niche drone services in high demand
- 4. Experienced management team with successful financial and technology background
- Disciplined acquisition strategy targets companies with advanced product offerings and specialized drone operational platforms and intellectual property

